The Effects of Weather on the Current and Future Ridership of Urban Cycling in North American Cities Considering a Changing Climate

By

Eduardo Adame Valenzuela

Department of Civil Engineering, McGill University, Montreal.

A thesis submitted to McGill University in partial fulfilment of the requirements of the degree of Master of Science (M.Sc.) Civil Engineering (Thesis) – Transportation.

©Eduardo Adame Valenzuela, 2023.

Table Of Contents

Table C	Of Contents2
Abstrac	ct5
Abrégé	
Acknow	vledgements7
Contrib	oution of the Authors
1. Ch	apter I: Introduction9
1.1	Context9
1.2	Objective of the Thesis
1.3	Structure of the Thesis 11
2. Ch	apter II: Literature Review
2.1	Cycling as a Transportation Mode12
2.2	Benefits of Cycling12
2.3	Cycling and its Promotion14
2.4	Cycling and Weather14
2.5	Cycling and Climate Change 16
2.6	Conclusions17
3. Ch	apter III: Analyzing the Behavior and Growth of Cycling in Four North
Americ	an Cities Before, During and After the COVID Pandemic
3.1	Abstract20
3.2	Introduction21
3.3	Literature Review
3.4	Methodology25
3.5	Data
3.5	5.1 Counters

	3.5	5.2	General Analysis	. 31
	3.5	5.3	Weather Variables	. 34
	3.5	5.4	COVID Variable	. 34
3	8.6	Re	sults	. 34
	3.6	5.1	Descriptive Analysis: Behavioral Patterns by Indexes	. 34
	3.6	6.2	Regression Analysis: Yearly Trends After Controlling for Weather	. 39
3	8.7	Co	nclusions	. 44
3	8.8	Ack	knowledgments	. 45
4.	Ch	apte	er IV: Bridge	. 46
5.	Ch	apte	er V: Effects of Climate Change on Future Yearly Cycling Demand for	the
City	y of I	Mon	treal	. 48
5	5.1	Abs	stract	. 50
5	5.2	Intr	oduction	. 51
5	5.3	Lite	erature Review	. 52
	5.3	8.1	Relationship Between Weather and Cycling	. 52
	5.3	8.2	Climate Change Models and Projections Downscaling	. 53
	5.3	8.3	Climate and Cyclists	. 55
	5.3	8.4	Machine Learning Methods and Cyclist Predictions	. 58
5	5.4	Me	thodology	. 58
	5.4	l.1	Definition of the Periods	. 58
	5.4	.2	Selection of the Sites and Cleaning the Data	. 59
	5.4	.3	Ridership-Weather Model	. 60
	5.4	4.4	Selection of the Climate Models and Downscaling	. 62
	5.4	.5	Prediction and Comparison	. 63
5	5.5	Re	sults	. 64

	5.5.1	Climate Projections Comparison	64
	5.5.2	Validation of the Daily Cyclist Counts Model	66
	5.5.3	Predictions for 2050	68
	5.5.4	Comparison Between 2019 and 2050	69
5	.6 Co	nclusion	74
5	.7 Acł	knowledgements	76
6.	Chapte	er VI: Discussion of the Findings	77
7.	Chapte	er VII: Conclusion and Summary	80
8.	Bibliog	raphy	82

Abstract

The presented work analyzes the influence of weather on urban cycling demand following two objectives: 1) to investigate the impact of weather conditions and bicycle demand using time-series models and 2) to carry out a climate analysis to estimate the cycling demand for 2050 under different climate model projections.

For the first objective, the proposed methodology characterized the temporal dynamics of the daily and weekly ridership profiles, identified the trends of utilitarian patterns, and determined the ridership trends across years while controlling for the influence of weather variations. Historical data of cyclist counts, and registered weather conditions for 12 years are used to calibrate models. In the second objective, the author estimated the annual profile of cyclist demand for 2050 based on the future climate projections following different emissions scenarios. For this purpose, an Extreme Gradient Boost (XGBoost) model was trained using historical counts and measured weather conditions to predict expected daily counts for the future year.

The results show a significant sensitivity to weather conditions and a constantly growing trend in cyclist volumes for Montreal that surged during the COVID-19 pandemic as new users adopted the bike for commuting or recreational uses. Regarding the future, predictions using different climate scenarios show a rise in bicycle demand in March, April, and November. However, the forecasts showed decreasing counts during the summer months due to extremely high temperatures drawing an annual reduction of cyclist demand in the high-emission scenarios. The results of this work could help plan and adapt better bicycle infrastructure and services considering a changing climate.

Abrégé

Le travail présenté analyse l'influence des conditions météorologiques sur la demande de cyclisme urbain en suivant deux objectifs : 1) étudier l'impact des conditions météorologiques et de la demande de vélos à l'aide de modèles de séries temporelles et 2) effectuer une analyse climatique pour estimer la demande de vélos en 2050 selon différentes projections de modèles climatiques.

Pour le premier objectif, la méthodologie proposée a caractérisé la dynamique temporelle des profils de fréquentation quotidiens et hebdomadaires, identifié les tendances des modèles utilitaires et déterminé les tendances de la fréquentation d'une année sur l'autre tout en contrôlant l'influence des variations météorologiques. Les données historiques des comptages de cyclistes et des conditions météorologiques enregistrées pendant 12 ans sont utilisées pour calibrer les modèles. Dans le deuxième objectif, l'auteur a estimé le profil annuel de la demande de cyclistes pour 2050 sur la base des projections climatiques futures suivant différents scénarios d'émissions. À cette fin, un modèle Extreme Gradient Boost (XGBoost) a été entraîné à l'aide des comptages historiques et des conditions météorologiques mesurées afin de prédire les comptages quotidiens attendus pour l'année à venir.

Les résultats montrent une sensibilité significative aux conditions météorologiques et une tendance à l'augmentation constante du nombre de cyclistes à Montréal, qui a bondi pendant la pandémie de COVID-19, de nouveaux utilisateurs ayant adopté le vélo pour leurs déplacements quotidiens ou leurs loisirs. En ce qui concerne l'avenir, les prévisions utilisant différents scénarios climatiques produisent une augmentation de la demande de bicyclettes au cours des mois de mars, avril et novembre. Cependant, les prévisions ont montré une diminution des comptages pendant les mois d'été en raison des températures extrêmement élevées, dessinant une réduction annuelle de la demande de cyclistes dans les scénarios à fortes émissions. Les résultats de ce travail pourraient aider à planifier et à adapter de meilleures infrastructures et de meilleurs services pour les cyclistes en tenant compte du changement climatique.

Acknowledgements

I wish to acknowledge the Mitacs Accelerate program for funding my research and to Eco-Counter Montreal where I carried out an internship during my thesis. Especially the Data Services group: Pierre Barban, Catherine Fralich and their leader David Beitel who helped me access the data and taught me more about data cleaning and management.

This project was funded in part by the Government of Canada. The Government of Canada's Environmental Damages Fund provided funding for this project under its Climate Action and Awareness Fund. This work was also supported by Fonds de recherche du Québec (FQRNT - Projets de recherche orientée en partenariat).

I am also thankful to all the members of the IMaTS lab at McGill University where I spent so much time working on the thesis, especially Léa, Daisy, Alejandro and Adham. Thank you for revising parts of the thesis or helping with specific questions. I also wish to thank the research laboratory leader, Professor Luis Fernando Miranda Moreno who helped me greatly not only as my supervisor in the writing and research of both articles but also in helping me to study at McGill and to expand my knowledge in the transportation field of civil engineering.

Thank you to Professor Van-Thanh-Van Nguyen who guided me as my cosupervisor in understanding the data, methodologies, and general knowledge regarding weather and climate models for both articles. I also wish to acknowledge Chenglong You from the Hydraulics lab who carried out the downscaling of the climate model to the city of Montreal for the second article.

I want to thank my roommates, Daniel, Cristina, and Daisy for uncountable moments of laughter, happiness, and incredible memories. Finally, I wish to say thanks to my family, Marcela, Gerardo, Ana, Lucía as well as Luis and our newest member, Elena. Thank you for your love, and for always being there for me.

Contribution of the Authors

The author's contribution to the first paper is as follows: study conception and design: D. Beitel, L.F. Miranda Moreno, V.T. Van-Nguyen; data collection: E. Adame Valenzuela, P. Barban; analysis and interpretation of results: E. Adame Valenzuela, P. Barban, D. Beitel, L.F. Miranda Moreno; draft manuscript preparation: E. Adame Valenzuela, D. Beitel, L.F. Miranda Moreno. All authors reviewed the results and approved the final version of the manuscript.

The contribution of the authors to the second paper is as follows: study conception and design: E. Adame Valenzuela, L.F. Miranda Moreno, V.T. Van-Nguyen; data collection: E. Adame Valenzuela, Chenglong, You; analysis and interpretation of results: E. Adame Valenzuela, L.F. Miranda Moreno; draft manuscript preparation: E. Adame Valenzuela, L.F. Miranda Moreno. All authors reviewed the results and approved the last version of the manuscript.

Introduction, literature review, Bridge, Discussion of the Finding, as well as the Conclusions and Summary of the thesis was written by the author, Eduardo Adame Valenzuela, under the supervision of Luis Fernando Miranda-Moreno and Van Than Van-Nguyen.

1. Chapter I: Introduction.

1.1 Context

Cities are spaces where people can access opportunities, and interact with each other in the search to improve their lives (1). Considering the exponential growth of cities in the past decades and their expected expansion in the future (2), investments in these regions provide a great opportunity to improve the well-being of a great number of people. In the last decades, this well-being has been challenged by problems related to air and noise pollution, safety, healthcare, and poor accessibility (3,4) that derive from the car-centric urban sprawl seen in most North American cities since the end of the second world war (5).

For these reasons, several cities have shifted their plans to include the promotion of more sustainable transportation modes. This transformation has caused a resurgence of the bicycle thanks to the vast investments to improve the infrastructure and the policies aimed at its promotion (6) which has derived economic, and sustainable benefits from bicycle use not only limited to the individual (7) but are also enjoyed by the community in general (7–10).

Nonetheless, cycling as a mode of transport does have some deterrents, one of them being its exposure to adverse weather conditions (e.g., high or low temperatures, precipitation, high humidity, and combinations of those weather factors). The literature agrees that there is a significant relationship between weather factors and cyclist demand (11–13) which only confirms what is evident for anybody who rides a bike. Generally, people do not bike on days that are too cold, too hot, raining, snowing etc. In the North American context, cities with cold winters such as Montreal have very noticeable bike seasons with demand peaking in the summer months and decreasing almost to a stop during the winter. Additionally, weather might offset the decline in cyclist trends (14) which could lead planners and researchers to formulate wrong conclusions when studying the impacts of certain policies or investments.

Although researchers and city authorities might have a good understanding of the seasonality and behaviour of the cyclist demand, global warming is expected to

disrupt the measured values of different weather variables in the future. Research seems to agree that, for the Canadian context, it is likely that the temperature will rise while the number of rainy days will decrease in the coming years (15–17). Additionally, the frequency of extreme weather events, such as heatwaves and flash floods, is expected to increase in the following decades. However, the effects of these changes on cyclist demand can not be generalized for different cities across the globe. This is because the direction and magnitude of the weather changes will vary for different regions and cyclists in every city will have dissimilar resilience to weather factors (18).

Hence, understanding the climate's effects on cyclist demand and how it might change in the future will help plan and design more sustainable and resilient transportation systems. This also includes the temporal impact assessment of strategies, projects and policies while controlling for the effects of weather or the identification of new infrastructure designs that allow people to cycle in adverse weather events (e.g., heat waves, winter, and high precipitation).

Despite the long literature on the impact of adverse weather on urban cycling, only a few works have looked at the long-term trends of cycling ridership while controlling for the variations caused by the weather conditions across seasons and years. Additionally, with the uncertainty in weather brought by climate change, there is no study, to the best of the author's knowledge, that tries to evaluate the changes in the cyclist demand in Canadian cities like Montreal under different climate-change scenarios. The motivation of this research is then to accurately understand the current and future trends of urban cycling considering the impact of climate and the expected variations under various climate scenarios.

1.2 Objective of the Thesis

Considering the past research on the relationship between weather and urban cycling, this work aims to investigate the temporal trends and relationship between historical climate conditions and urban cycling to better understand the present and future trends of cycling under alternative climate scenarios.

More specifically, the objectives of this work are:

- To propose a methodology that characterizes the historical temporal dynamics of utilitarian and recreational cycling across the years before and after the COVID-19 pandemic while controlling for weather factors. The weather-controlled cycling volume growth trends are determined for four North American cities as a case study.
- 2) To propose a methodology to predict future cycling ridership under alternative climate scenarios combining cycling demand models as a function of weather conditions and climate-model outcomes. Projections of cyclist demand for the year 2050 are made using the city of Montreal as a case study.

1.3 Structure of the Thesis

Chapter II presents a general literature review of the context of urban cycling, its relationship to the weather, and the challenges of climate change. Chapter III contains the research following the first specific objectives, which resulted in an issued article and is presented as it was published in the Transportation Research Record 2023. Chapter IV provides a general review of the first objective and links the discoveries made with the second objective. Chapter V presents the research that followed the second specific objective, which will be presented to be published as a second article and is shown as the author intends to submit it. Chapter VI encompasses a general discussion of the findings regarding the previous chapters. Chapter VII provides the conclusion and a general summary of the thesis.

2. Chapter II: Literature Review.

2.1 Cycling as a Transportation Mode

Cycling as a transportation mode is no novelty at all. A cycling boom ploughed the American cities of the 1890s as people started using bicycles more and more for commuting and leisure trips, in part thanks to their characteristic as private vehicles that could take you wherever they wanted without the limitations of timetables and fixed routes of the streetcars (19). Nonetheless, after walking in any city in the United States or Canada it is easy to realize this is not the case anymore. Since the 1950's high levels of motorization, city sprawl, car-centric infrastructure, and government policies that promote the automobile have caused a high adoption of private motorized vehicles and a sharp decline in cyclists (20). Even so, if the bicycle does not have the same modal share as before, there is still a percentage of the population that continues to cycle in several North American cities.

2.2 Benefits of Cycling

This continuous use of the bicycle by individuals is due to the many benefits it provides to them. First, due to the low skill required to bike, it is a transportation mode available to a large segment of the population (21). In 2013, 82% of Canadians aged between 12 to 14 years old reported cycling in the previous year (22). Second, it can be competitive against the use of cars for short trips. Ellison and Greaves (23) compared trips from 178 motorists in Sydney, Australia and concluded that 90% of the trips below 5 km would only take up to 10 minutes more for an inexperienced cyclist. Third, it provides economic benefits to the individual being less expensive compared to a private vehicle and even to public transport (24), having a private cost per kilometer 6 times lower than the car (25). Fourth, commuting using a bicycle provides several health benefits. Following a literature review, de Hartog et al. (26) concluded that shifting the commuting mode from car to bicycle would gain 3 to 14 months in life expectancy and concluded that the benefits from the activity outweighed the negativities of exposure to pollution and the risk of accidents. The moderate 30-minute physical, which can be achieved by travelling to work daily on

a bike, can lead to a 19% reduction in the mortality risk (21). People who frequently cycle have reported lower levels of psychological distress and higher levels of life satisfaction (27), which remarks the additional mental health benefits of commuting by bike.

The use of the bicycle as a transportation mode does not only provide benefits for individuals but also the cities. First, it requires a lower amount of public space dedicated to that mode. For example, cyclists in Rajkot City, India, consumed 12.5% of the space-time required by a car (28) while, in France, a cyclist needs 19% of the space-time consumed by a car-driver (29). Both results confirm that bicycles are more space efficient than private cars implying that a modal shift from a significant part of the population would liberate public areas that could be used for other functions such as greenery, urban furniture or increasing development. The lower space consumption can also be related to congestion levels. This is reflected in Copenhagen, where cyclists are presumed to represent a lower cost for congestion to the city (30). This is in line with the findings in Washington D.C. where bike-sharing stations were negatively related to congestion levels (31). In regards to health, Bassett et al. (32) found a negative relationship between active mobility modal share and obesity levels. Gotschi estimated that, for Portland, Oregon, 30 minutes of daily biking saved \$544 annually per person to the healthcare system (33). In general, the promotion of active mobility networks has beneficial effects on society when considering road safety, parking, and health costs (34) along with congestion, infrastructure maintenance, pollution, and quality of life (25). In Norway, Lanus et al. (35) deemed that a Cycle Investment Model would be highly cost-effective based on the gains of quality-adjusted life years (in monetary terms) it would achieve. The literature confirms that investing in cycling produces benefits for the cities, giving a strong argument for the expansion of bike facilities and the promotion of this mode.

Although the literature has given plenty of reasons to invest in cycling, in the last decades there is another one that has spurred into debate. The concept of sustainability has gained significant momentum within the research, political and community agenda (36). With the commitment of countries to reduce humanity's

impact on the environment, cities have been identified as strategic locations for the reduction of greenhouse gases in the world, since they host a significant proportion of the population and enterprises with major world production contributions, which would be at significant risk from the effects of climate change (37). These factors place the urban areas as key players in the mitigation of emissions and transition to a low-carbon lifestyle (38). In recognition of the situation, cities have designed plans that will help guide their development, in which cycling and walking could work as zero-carbon sustainable solutions for urban transportation (39). For example, a study in Adelaide, Australia, found that a 5 to 40% shift in Vehicle Kilometer Travelled (VKT) from private vehicles to alternative transportation modes such as cycling, would result in 0.15 to 0.95 million fewer tons of CO2 in 2023 compared to a business-as-usual scenario (40). Because of these results along with a desire to fulfil their sustainability mode, urban areas have started to adopt cycling-related strategies such as the installation of bike-sharing systems to reduce greenhousegas emissions, although the magnitude of the reduction will depend on the modal shift (41-44).

2.3 Cycling and its Promotion

For all the benefits and reasons stated above, cities across the globe are promoting the use of bicycles through investment in infrastructure and the passing of policies even before state or national legislatures (45). In 2016, bike-sharing systems had expanded to 850 from a handful in the 1990s, which shows the refocusing of urban transport planning towards more sustainable modes (46). In American cities, bike trips doubled from 1977 to 1995 (6) while in Canadian cities, a study found a ten percent growth from 1996 to 2016 (47). This interest in cycling shows a need to further the research for its promotion, regarding the valid strategies, the impact and, maybe more importantly, the challenges it will face.

2.4 Cycling and Weather

One of the main challenges related to the use of the bicycle for commuting trips is the exposure of the mode to weather factors which has been identified as one of the discouraging factors(48), having an important impact on the overall travel experience (49). Due to its major influence, the weather has been proven to be an important predictor that can improve the estimations of the Annual Average Daily Bicyclists Traffic (AADBT) (50).

The research regarding the relationship between daily travel behaviours and weather conditions is quite extensive. Böcker et al. (51) carried out a systematic literature review of 54 published articles and concluded that warm and dry weather favours the use of active transportation modes while rain, snow, and hot weather conditions cause a shift of users to sheltered modes and can also lead to the cancellation of trips with outdoor destinations. From the review, they concluded that warm and dry days affected active traveler volumes positively while rain, wind, and hot days affected them negatively however the relationship does not seem to be linear. The magnitude of the effects also depended on the region the paper was carried out suggesting that people living in different climates will present different levels of resilience against the specific weather factors. In their overview of the literature focusing on the main factors affecting urban cycling Heinen et al. (52) dedicated a whole section to the effects of weather, acknowledging that variations in the day-to-day conditions could affect cyclists' daily decisions. Their overview showed that precipitation harmed cyclist counts, however, there was no clear consensus regarding its magnitude while temperature was deemed to affect cyclist counts positively. They also noted that commuters seem to be less affected by the weather conditions compared to other riders since they can not cancel nor modify their trip. Regarding the Canadian context, the highest levels of activity in Toronto's Bike Sharing system were associated with perceived temperatures between 20 and 30°C, while rain, snow, and humidity harmed cycling activity (53). In Montreal, Miranda-Moreno and Nosal (54) concluded that temperature increases will result in greater bicycle ridership up to a certain point, while humidity reduces them. As expected, rain reduced the cyclist in the same hour but also presented lagged effects up to three hours. Additionally, they noticed that users will respond differently to the weather depending on the season, for example, extreme increases in temperature would promote cycling in winter and reduce them in summer.

2.5 Cycling and Climate Change

In the present decade, cities have recognized the new challenges brought by global warming and its effects can already be felt in the world, with more frequent and greater weather events already causing losses in nature and human societies (55). It is expected that the number of hot days, heatwaves, and intense precipitation events are very likely to increase, while decreases in very cold days are likely to occur (56). In the Canadian context the number of days with a maximum temperature greater than 25°C, 30°C, and nights with maximum temperatures greater than 22°C has already increased from 1946 to 2016 in the southern regions, which is derived in warmer winters and summers (57). Canada's Changing Climate Report (58) estimates an increase in temperature during the last and in the coming years at double the mean of the projected global average. Regarding precipitation, the report estimates an increase in precipitation during the winter and a decrease during the summer with an overall annual growth.

Regarding transportation, climate change will likely impact the sector. In 2008, Koetse and Rietveld (59) carried out a literature review to understand the impact climate change would have on transportation and concluded that the topics had received little attention in research, however, one could expect the transportation systems to perform worse under extreme conditions. Considering the impacts of climate change on transportation systems is critical since projects could fail to meet their objectives due to a shortening of their useful lifespan caused by evolving travel patterns or because of disruptions from extreme climate events (60). While some authors argue that the extreme events brought by the warming climate may interrupt the functionality of urban transportation networks (61) others have suggested that climate change will modify the annual patterns of cyclists and pedestrians, with cities with colder climates, such as Montreal, possibly gaining active mobility users (51). The exact impacts of climate change are unknown and whether these experiences are positive or negative for urban cycling will likely vary for each city.

2.6 Conclusions

Although the literature review has confirmed the relationship between the weather and cyclist demands, there are a few gaps in the literature:

- Few studies have analyzed the growth of cyclist volumes in North American Cities while controlling for the effects of climate and the impacts brought by the COVID-19 pandemic.
- There are few studies regarding the possible variation in future cyclist demand in Canadian cities considering the effects of climate change under different emissions scenarios.

Chapters three and five showcase two methodologies that tackle these gaps in the literature and provide empirical evidence by applying them to data from cities in North America as case studies. Chapter III: Analyzing the Behavior and Growth of Cycling in Four North American Cities Before, During and After the COVID Pandemic

Eduardo Adame Valenzuela

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: eduardo.adamevalenzuela@mail.mcgill.ca

Pierre Barban

Data Services

Eco-Counter Canada/USA

604-3981 St-Laurent Boulevard, Montreal, Canada H2W1Y5

Email: pierre.barban@eco-counter.com

David Beitel

Data Services

Eco-Counter Canada/USA

604-3981 St-Laurent Boulevard, Montreal, Canada H2W1Y5

Email: david.beitel@eco-counter.com

Luis Fernando Miranda Moreno

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: luis.miranda-moreno@mcgill.ca

Van-Thanh-Van Nguyen

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: van.tv.nguyen@mcgill.ca

Key Words: Cyclist Demand, Weather, Pandemic, Trends, Regression.

Submitted 19th January 2023

Data Statement: The cyclist counts data used for this study is property of the city of New York, the city of Montreal, the city of Ottawa, and the city of Vancouver. Any request for the cycling counts data must be submitted to the corresponding authorities or information portals of each city. All the weather measurement data were acquired from World Weather Online.

Funding Statement: The collaboration between McGill University and Eco-Counters that produced this article was funded by the Mitacs Accelerate program.

3.1 Abstract

The paper highlights the changes in cycling patterns and ridership trends across 12 years (including the COVID-19 pandemic) in Montreal, Vancouver, Ottawa, and New York. Using data from 17 bicycle counting stations, changes in the dynamics of daily and weekly profiles before and during the pandemic were determined. Additionally, the ridership demand evolution across the years is explored using models that control for the variations in the weather. All the studied bicycle facilities experienced a change in the daily and hourly patterns in 2020 (the first year of the pandemic), tending towards recreational purposes. Buehler and Pucher (62) had previously found a significant growth in bicycle activity during the first year of the pandemic, but the trends for the following years (2021 and 2022) have not been studied. This paper found that all counting sites located on cycling facilities primarily used for utilitarian purposes experienced a growth in ridership during 2020. Ridership on utilitarian corridors in Montreal and New York City grew considerably during the pandemic before stabilizing in 2021 and 2022. The same counting sites returned quickly to utilitarian hourly and daily patterns in 2021. The mixed-utilitarian bicycle facilities in Ottawa and Montreal shifted towards more recreational uses during the pandemic, though ridership did not grow in 2021 and 2022. All the counting sites in Vancouver shifted towards mixed-use during the first year of the pandemic and have not shown any clear sign of returning to utilitarian patterns.

3.2 Introduction

During the last decades, many North American cities have seen an increase in bicycle ridership, which in part can be related to the investment in cycling infrastructure (6). With the energy and climate change concerns, cycling is seen as one of the most sustainable transportation modes. Cycling not only reduces fossil fuel consumption and related emissions but also helps improve public health and well-being (63). Bicycle trips represent a relatively small modal share compared to automobiles in North American cities; however, an important increase has been observed in many cities. The COVID-19 pandemic in 2020, had a major impact on transportation demand. While public transportation mode share reduced significantly, bicycle mode share growth became more pronounced (62).

During the pandemic, bicycle infrastructure played a critical role in cities, offering an alternative transportation mode for commuters with public health and safety concerns related to public transit systems. A positive correlation between bicycling sharing system (BSS) trips and the number of new daily COVID cases in New York City, as shown by Teixeira and Lopes (64), suggests a possible modal shift from the subway system toward the CityBike system. Bicycle infrastructure also provided the general population the opportunity to safely participate in recreational activities during the pandemic (65). Bicycle infrastructure helped ensure access to destinations and activities during the pandemic with minimum economic and environmental impacts. Accordingly, cycling has emerged not only as a sustainable but also as a resilient mode of transportation (66,67).

Given the important role of cycling as a sustainable mode of transportation, an important body of literature has been published in the last few years. A large body of research has documented key issues that limit cycling participation such as road safety, the lack of infrastructure, or the impact of weather (68,69). A few papers have also looked at the cycling ridership evolution before and during the pandemic (70,71). Most research explored the impact of weather before the pandemic or the challenges that the pandemic posed to bike-sharing systems – with significant changes in the usage patterns of share-mobility bicycle services (64,66,72–76).

21

These works are important in that they study the resilience of bike share systems. However, they do not reflect the entire picture of the bicycle infrastructure usage and bicycle demand during these critical pandemic years. Buehler and Pucher investigated the impact of the pandemic on cycling in various cities and countries in Europe, North America and Australia (62). Using data from 2019 and 2020, they attempt to establish the overall trends as well as the variations over time.

Despite the significant amount of work in the last years, to our knowledge, very little work has explored the long-term trends and detailed patterns before and during the pandemic in various cities in North America. As reported by Buehler and Pucher, a significant growth in bicycle activity was observed in the first year of the pandemic, but there is less knowledge of trends in 2021 and 2022. Additionally, past research on cycling trends has not controlled for the potential impact that weather may have in the seasonal-yearly variations, leading to results that under- or over-estimate cycling growth.

This paper illustrates the long-term bicycle patterns and trends before and during the pandemic across four major cities in North America (Montreal, Vancouver, Ottawa in Canada, and New York City in the United States) during the 2010-2022 period. The specific objectives are: i) to propose a methodology to characterize the temporal dynamics of the daily and weekly ridership profiles to identify the trends of utilitarian patterns and ii) to investigate the ridership trends across seasons and years in four cities while controlling for weather variations. Understanding the behaviour and the growth trends occurring in each city may provide city planners and authorities with greater knowledge of current and future cyclist demand. This knowledge can be used to inform modifications to existing facilities and the creation of new active transportation routes. Additionally, controlling for weather allows for a fair comparison of cities located in different climates.

The following sections in this paper will provide a review of the existing literature, a summary of the methodology, a description of the cities and the data, and an analysis of the results to better understand the temporal variations of cyclist behavior and demand in the different cities.

3.3 Literature Review

Trying to understand the changes in travel behavior by analyzing the census data in Seoul's Metropolitan area, Choi et al. discovered that bike trips increased in duration and frequency from 2002 to 2006 (77). Pucher, Buehler, Merom, et al. stated that cycling levels stagnated in the United States from 2001 to 2009 based on the results of the National Household Travel Surveys (78). When analyzing the trends in demand for London's bike-sharing system Chibwe et al. found that the number of trips increased throughout the years as the unemployment rate decreased, which could mean that most of the scheme's trips are of a utilitarian nature (71). Ledsham et al. analyzed the results of surveys by using generalized structural equation models to better understand the factors influencing the amount of utilitarian and recreational trips in the suburbs of Toronto (79). Stinson et al. created statistical models capable of estimating the amount of utilitarian and recreational trips in the city of Los Angeles. Nonetheless, the author of this paper could not find any research regarding temporal changes in usage patterns of bike activity throughout the years (80).

Nankervis, in 1995 compared the daily counts of cyclists in an Australian university to a weather condition index, which was made up of the force of the wind, the maximum daily temperature, and the presence of rain (11). Although he did find a relationship between the variables, the magnitude of the influence was weaker than first assumed. However, the author mentions the study focused on the behavior of students which may not be representative of the general population. This change in commuting behavior may be explained by the change in the perception of safety cyclists experience in adverse weather conditions (12). On an interesting note, the behavior could be affected not only by the present meteorological conditions but also by the weather forecast (81). By applying a log-linear regression model on automatic bike counts, Thomas et al. stated that temperature had the greatest influence on the counts while wind speed, hours of sunshine and precipitation did have a significant effect and also found that favorable weather could hide a downward trend in bicycle usage (24). Miranda-Moreno and Nosal noted that temperature had an important positive effect and humidity harmed bike counts, while precipitation could have a

significant lagged negative effect of up to three hours on the observed counts. They also discovered that above a certain threshold cyclists' desire to bike starts to decline with higher temperatures. The threshold was calculated at 28C, 28C, 30C and 27C in Montreal, NYC, Seattle, and Austin, respectively (54,82-84). This indicates that the magnitude of the effects of the weather conditions on cyclist demand will vary from city to city, as also noted by Goldmann and Wessel (18). The magnitude of the effect also varies depending on the use of the bike route. Zhao et al. found that weekday users of a bike train in Seattle were more resilient to weather conditions than their weekend counterparts (27). Also in Seattle, Niemeier discovered a greater variability of counts in the afternoon peak (from 3:30 pm to 6:00 pm) and stated that this could be partially attributed to the presence of non-utility cycling trips (85). Miranda-Moreno and Nosal in their study of the data from automatic counters in Montreal found that recreational locations were more sensitive to weather (25). Wessel also found that the effect of lighting conditions differs between utilitarian, recreational, and mixed-type users (31). Pazdan et al. also carried out the classification of nine different bicycle corridors in the city of Krakow (32). Then they added the site's classification as an independent categorical variable in a regression model of the bike counts which turned out to be not significant. When determining the influence of weather variables, studies have used regression models such as linear and log-linear (14,18), square root (87), log-linear in absolute and relative models (54) and Negative Binomial with log identity (81,88). While those models dealt with the non-linear relationship of temperature and by using squared terms and of precipitation by categorizing that variable, other studies have opted for the use of General Additive Models (82,84,88) which, instead of assigning a constant coefficient, the influence of each variable is determined by a function. When dealing with time series data, it is important to control for the autoregressive temporal effects (83). Some studies have addressed this issue by using auto-lagged effects (13,82) and by transforming the data using a 9-term average of the counts (83,89). Others have studied the use of Auto-Regressive Integrated Moving Average (ARIMA) models to account for this temporal correlation between observations (87,90).

Regarding the effects of the pandemic on cycling demand, Buehler & Pucher investigated the overall trends in different cities across, Europe, the Americas, as well as Australia by using automatic counter data. The authors concluded that cycling increased from 2019 to 2020 in most of the studied cities (3). Nikiforiadis et al. investigated the impacts of the pandemic on the perception of bike-sharing systems (BSS) in the city of Thessaloniki in Greece. A field survey found that the pandemic made the city's BSS a more attractive option and could become the preferred commuting mode for current car passengers and users already registered in the system (76). In London, Heydari et al. compared the observed cycle hires in the city's BSS with the estimated counts if the pandemic had not occurred. They found that, although counts initially dropped in the first months, they rapidly bounced back to the expected levels which is a strong indication of the system's resiliency (75). The authors also noticed an increase in the duration of trips which could be caused by a shift from public transit users to the BSS. Similarly, Shang et al. also discovered an increase in the average duration of bike-shared trips in Beijing during the pandemic (91). Wang & Noland, analyzed data from New York City's subway and bike-sharing systems, comparing the daily counts from 2019 to 2020 while controlling the effect of weather. They concluded that both systems saw an initial decrease in riders during the first months of the pandemic. However, by September 2020, the BSS had nearly recovered to pre-pandemic levels while the subway's rider counts remained low (67). Nguyen & Pojani found, through a face-to-face survey, an increase in recreational cycling in the city of Hanoi, Vietnam. They also stated that income and age were not significant in the choice of taking up more recreational trips and that most people adopted cycling as a way to increase or maintain their level of physical activity and to socialize in an infection-safe environment (92).

3.4 Methodology

The proposed methodology consists of six steps as shown in Figure 1:



Figure 1: Methodology flow chart

The first step is the selection of bicycle counting sites to be used in this research. The selection process is based on the following criteria: i) the counters should be located on main corridors with few, or no alternative routes; ii) the counters should have historic long-term data of at least 8 years of relatively complete counts; iii) the facilities should exhibit a mainly utilitarian pattern in the years preceding the pandemic; iv) the counter should be located on corridors with high annual average daily counts.

As a second step, each counting site was classified per day as utilitarian, recreational or mixed based on their daily and weekly patterns by using the indexes defined in [1] and [2], which were developed in a similar way to the ones presented by Miranda-Moreno et al. (93). The hourly index is defined as follow:

$$I_{AM/Noon} = \frac{V_{peak}}{V_{out}} = \frac{\frac{1}{2}\sum_{i=7}^{8}V_i}{\frac{1}{2}\sum_{i=11}^{12}V_i} \dots [1]$$

Where $I_{AM/Noon}$ is the hourly index computed as the ratio of the V_{peak} or the average peak-hour volume from 7:00 a.m. to 9:00 a.m. and V_{out} , the average off-peak volume from 11:00 a.m. to 1:00 p.m. Here, V_i stands for the number of counts at the hour *i*,

and *n* is the number of hours in the period (2 hours for the AM peak from 7:00 a.m. to 9:00 a.m. and 2 hours for the noon peak going from 11:00 am to 1:00 p.m.).

The weekday versus weekend is defined as:

$$I_{wd/we} = \frac{D_{wday}}{D_{weekend}} = \frac{\frac{1}{5}\sum_{i=1}^{5}D_i}{\frac{1}{2}\sum_{i=6}^{7}D_i} \dots [2]$$

Where $I_{wd/we}$ is the daily index computed as the ratio of the D_{wday} or the average daily volume from Monday to Friday (i = 1 on Monday, i = 2 on Tuesday, etc.), and $D_{weekend}$, which is the average daily volume from Saturday to Sunday. These formulas would give a unique am/noon index for each day on the historic data, excluding weekends, and the same weekday/weekend index for every day on the same week.

As a third step, the days were classified using a similar process to the one presented by Wessel (81). If both indexes were greater than one, the day would be considered to have a utilitarian pattern, if both indexes were less than one it would be considered as recreational, and as mixed for every other case, that is:

If
$$I_{AM/Noon} > 1$$
 and $I_{wd/we} > 1$ then day type = Utilitarian.

If $I_{AM/Noon} < 1$ and $I_{wd/we} < 1$ then day type = Recreational.

Then, for every year, the percentage of utilitarian, recreational, and mixed days was calculated to create the line graphs presented in the Results section.

In the fourth step, all counting sites are classified into four possible groups, the same classifications used by Miranda-Moreno et al. (93). If the counting site has 60% or more of its days classified as utilitarian, then the site is considered to be used for purely utilitarian purposes. If 60% or more of the days are recreational, then the site is classified as purely recreational. All the rest are classified as mixed and divided into two groups: mixed-utilitarian when the number of utilitarian days is greater than the number of recreational days, or mixed-recreational in the inverse case, that is:

If *utilitarian days* >= 60% of all days in the year, then *site type* = Utilitarian.

If *recreational days* >= 60% of all days in the year, then *site type* = Recreational. Else:

If utilitarian days > recreational days, then site type = Mixed - Utilitarian

If utilitarian days < recreational days, then site type = Mixed - Recreational

Afterwards, for all counting data from the same city and of the same type (for example Montreal-Utilitarian, Montreal-Mixed, etc.) their total daily counts were paired with the daily measured weather factors, the daily stringency index, a holiday dummy variable (1 if the day is a national holiday, 0 if not), a bridge dummy variable (1 if the day after a holiday or a Monday before a holiday, 0 if not), as well as with a location variable (site id), and to the year the count was taken. Similar weather variables, such as maximum daily temperature and average daily temperature, were compared and those with the highest correlation to the counts, or with the greater explanatory power, were kept for future analysis. Afterwards, another correlation analysis was carried out to ensure the independence of the explanatory variables. If there was a high correlation between the two of them (equal or greater than 0.5), the variable with the greater correlation to the counts was kept.

In the fifth step, a regression analysis was carried out by fitting each data-frame, optimized to eliminate any correlated or nonsignificant variables, to two different regression models wildly used in similar studies: log-linear and the negative binomial with log identity.

In the last step, the yearly coefficients of the models were analyzed to determine if the cities were experiencing an overall growth in cycling demand independent of weather conditions.

3.5 Data 3.5.1 Counters

The study used bike counts from counters manufactured by Eco-Counter and installed in cities with an open data access policy. The study only used counts

classified as "bike" by the counter algorithm. A first selection was made by identifying the counting locations on cycling routes with few (or no) alternatives. Therefore, counting locations on bridges, underpasses and major cycling routes were prioritized. This reduced the likelihood that pop-up bike lanes, or new infrastructure in general, would impact growth trends at each counting location. From the locations selected in the first stage, only the counters that showed a small number of outliers in their historical data were selected. Additionally, all locations with a low mean daily count compared to the other sites of the same city were dropped. This generated a total of 17 sites to be used in the study. The extreme values from these sites were identified using a visual time-series analysis and manually replaced by a null value. For the cases where three consecutive days or less had null values, the counts were linearly interpolated. All other null values were deleted from the analysis. Additionally, there were some additional deletions of data for specific counters which will be explained in the following paragraphs.

The data spans from the installation of the counter until June 30, 2022. Table 1 shows the summary statistics for all the daily cyclist counts collected at each of the studied counters. Although the number of total sites per city might be smaller than the ideal, this allowed a more relevant dataset and a careful inspection of the counts. However, the low number of studied sites remains a limitation of this paper.

3.5.1.1 New York City

New York is the most populous city in this paper with a population of 8,550,405 in 2015 and a density of 10,474.7 people per square kilometer (94). The city experiences warm and humid summers followed by mild to cold winters (95). By 2018 the city had installed 1,240 miles of bike lanes and had a daily average of 490,000 cycling trips. In 2022 the borough of Manhattan scored 54 over 100 according to the City Rating made by the People for Bikes Organization (96), giving the area the 11th place within the ranking of large cities. The selected counters were the Williamsburg Bridge Path (NY1), the Ed Koch Queensboro Bridge Shared Path (NY2), and the Manhattan Bridge Display Bike Counter (NY3). The bike counts from the Manhattan Bridge pedestrian path were added to the NY3 site to account for any

bikers that may have changed routes because of the partial or total closure of the cycling path.

3.5.1.2 Montreal

In 2021 Montreal had a population of 4,291,732 and a density of 919 residents per square kilometre (97). Currently, there are 889 kilometers of bike lanes in the city (98). In City Ratings, Montreal scored 65 points over 100, making it 1st place among the large cities of North America (96). The selected counters are located on the Jacques-Cartier Bridge (MT1), Rue Rachel near the Papineau intersection (MT2), Boulevard Maisonneuve near the intersection with Rue Peel (MT3), Côte Sainte-Catherine near the intersection with Rue Stuart (MT4), Avenue du Parc near the intersection with Rue Duluth (MT5), and Rue Berri near the intersection with Rue Ontario (MT6). The 2020 counts of the MT6 site were deleted due to a construction project that reduced access to Rue Berri.

3.5.1.3 Ottawa

The Ottawa metropolitan area had an estimated population of 1,135,014 and a population density per square kilometre of 243.3 in 2021 (97) which makes it the smallest and least dense city of this study. The city scored 51 over 100 in the People for Bikes City Rankings of 2022, placing it as the 13th best place to bike between the large cities of North America (96). The selected counters were the NCC Eastern Canal Pathway Colonel By (OT1), one on Avenue Laurier near the intersection with Metcalfe (OT2), the Trillium Bayview (OT3), the Trillium Gladstone (OT4), and the NCC Alexandra Bridge Cycle Track (OT5).

3.5.1.4 Vancouver

The metropolitan area of Vancouver had a population of 2,642,825 and a population density per square kilometre of 918 in 2021 (97). With a cycling network of 325 kilometers in 2018 (46) the city scored 56 points out of 100 in the 2022 City Rankings (96), which gave it the 6th place of all large cities. The selected counters are located

on the Seawall path near Science World (VN1), on the Seawall path near the Creekside Community Centre (VN2), and Burrard Street near the intersection with Cornwall Avenue (VN3). Due to the presence of abnormal counts during 2012, that year of data was deleted from the VN1 site.

Site	Mean	Median	Maximum	Minimum	Standard deviation				
New York City									
NY1	4830	4820	10940	162	2461				
NY2	3702	3689	8295	21	1793				
NY3	3953	3907	9466	117	2083				
Montreal									
MT1	1138	835	5857	0	1123				
MT2	2725	2325	9772	0	2200				
MT3	2696	2174	11092	0	2358				
MT4	1133	930	5337	0	1015				
MT5	1451	1147	5305	0	1315				
MT6	2347	1805	8812	0	2101				
		Ot	tawa						
OT1	831	619	3340	0	774				
OT2	994	650	4128	2	948				
OT3	427	272	2350	0	471				
OT4	528	390	2724	0	495				
OT5	931	751	3284	0	866				
Vancouver									
VN1	3622	3151	10187	0	2272				
VN2	1973	1440	8823	0	1716				
VN3	3585	3245	10129	56	2084				

Table 1: Summary statistics of the daily cyclist counts for each counter (units in cyclists per day).

3.5.2 General Analysis

Analyzing the boxplots presented in *Figure 2*, the New York sites' median values for daily cyclist counts do seem to have increased slightly in the last years. Looking at

the Montreal counters, there is no visible trend citywide, which is a similar case in Vancouver. Most of the counters in Ottawa seem to show a downward trend in the last years, even before the start of the lockdown measures. However, it would be wrong to assume, based on these plots, that cycling demand in the cities has decreased in the last years. As noted in the literature, it is important to control for any external variables, such as the general weather and the lockdown measures that could hide the inherent trend of cycling demand before making any assessment about future investments.



Montreal







Figure 2: Boxplot of the daily cyclist counts from each of the studied sites (units in cyclist per day)

3.5.3 Weather Variables

The weather data was obtained through the *World Weather Online*'s API. For each city, the measured weather data was recovered from the station closest to the centroid of all the counters in that city. The initial weather variables extracted for analysis were daily maximum and average temperature (degrees Celsius), daily average and maximum wind speed (kilometers per hour), daily total precipitation (millimetres), daily average and maximum precipitation intensity (millimetres per hour), daily average and maximum humidity (percentage), as well as the daily percentage of the sky that was obscured by clouds, also known as cloud cover.

3.5.4 COVID Variable

Ritchie et al. (2020) calculated per day a *stringency index* which is the mean score of nine different metrics: school closures, workplace closures, cancellation of public events, restrictions on public gatherings, closures of public transport, stay-at-home requirements, public information campaigns, restrictions on internal movements, and international travel controls. The possible values have a range from 0 to 100 (where 0 means there are no restrictive measures in place and 100 means the strictest response by the authorities).

3.6 Results

3.6.1 Descriptive Analysis: Behavioral Patterns by Indexes

From *Figure 3* it is observed that most of the selected bicycle corridors show a utilitarian pattern before the start of the pandemic. During the pandemic (2019-2022), the studied facilities experienced a shift towards more mixed or even fully recreational patterns. This is expected as the lockdown measures required people to work from home, which diminished the number of commuter trips. The biggest impact can be seen in 2020 when six of the seventeen sites shifted from utilitarian to mixed and three others shifted from utilitarian to recreational. However, the last two years (2021 and 2022) show a recovery trend toward mixed-utilitarian or purely utilitarian patterns in most cases, which could suggest an eventual return to previous uses in most of the study corridors.



Figure 3: Temporal changes in the usage of the facilities

Two of the New York counting sites had some of the biggest declines in utilitarian days during 2020. Nonetheless, all three of them showed a fast recovery towards the patterns observed before the start of the lockdown measures. Ottawa was another city with sharp declines in utilitarian days during the pandemic. However, unlike New York City, the recovery towards past patterns occurs more slowly in three of the sites. In the other two, both located on the Trillium path, the change toward mixed patterns seems to be permanent as the number of utilitarian days stagnated after 2020. Ottawa also registered the highest percentage of recreational days compared to the other study cities.

Montreal was the city with the most consistent patterns throughout the pandemic. Although all the locations showed a dip in 2020, most of the locations still mainly experienced utilitarian days. Only two of them had an important shift in 2020 towards mixed patterns. These graphs are more closely related to those from New York City, although with less sharp 2020 dips as seen in *Figure 4*.

Vancouver showed a different behavior, being the only city with more mixed pattern days in all its counting sites. It is important to note that the VN2 location showed a general trend towards more recreational use before the start of the pandemic. The other two sites were like the ones seen in Ottawa, where utilitarian patterns shifted towards mixed and did not show any sign of shifting back. In both, there is a significant decline of utilitarian days in 2020, then a small increase in 2021, followed by a smaller dip in 2022. It is interesting to see the indexes in this case, as shown in *Figure 5*, Vancouver is the only city where the weekday/weekend ratio continued to decline in 2022 even though the AM/noon peaks ratio is increasing slightly. This mismatch in tendencies explains the mixed classification.




3.6.2 Regression Analysis: Yearly Trends After Controlling for Weather

A regression analysis was carried out for every type of route in the four cities. Each data frame was optimized by eliminating all correlated variables and by only keeping significant variables at a 95% confidence level. Due to space limitations, only the coefficients and results for the Negative Binomial models of the utilitarian facilities are shown in *Table 2* since their R² score showed a lower variability. The coefficients calculated by the models, which were built with R functions, were graphed as the percentage change in counts through time compared to a base year, which was the first year of available data for every individual dataset.

The utilitarian bike facilities located in the cities of New York (NY1, NY2, and NY3) and Montreal (MT2, MT3, MT4, MT5, and MT6) show a consistent growth throughout the years, which increased in the first year of the pandemic. This could be attributed to the implementation of "pop-up" bike lanes, the closure of streets, or the shift of public transport users towards biking due to the loss in the level of transit service and greater concern of infection in crowded vehicles. The trend seems to stabilize in 2021 and 2022 for both cities which could be a sign of the general recovery of service in the main public transport services in those cities. Ottawa's utilitarian facilities (OT1, OT2, OT4, and OT5) have also experienced consistent growth throughout the years. However, for the first half of the timeframe they showed lower levels of demand compared to the base year.

The Mixed-Utilitarian cycling corridors in Ottawa (OT3) and Montreal (MT1) show a different story with a more erratic growth pattern than their utilitarian counterparts. Both show a downward trend in demand which was magnified during the first year of the pandemic. In the last years, the decrease seems to have stagnated in Ottawa and even reversed in Montreal. It must be noted that the data analyzed came from one counter for both cases which means that this behavior represents only a local situation and cannot be assumed for the rest of mixed-utilitarian sites in the cities. Future research should be carried out to confirm if this is a city-wide or simply a local trend.

All of Vancouver's locations were classified as a different type. Site VN3 was the only utilitarian route in the analysis and showed a stagnated growth except in 2017 when it suffered a decrease in counts. As with the other utilitarian sites, the growth increased in 2020. Nonetheless, it was not capable of retaining the higher demand in the following years. The site VN1 was classified as Mixed-Utilitarian and shows a similar trend to the one observed in the utilitarian facilities with constant growth throughout the years, except for 2017, and stagnation in the last two years. As with the mixed utilitarian corridors, these results may be only representative of the location of the counter and may not be assumed as the general situation of the city.

It is important to note that the models of the mixed-patterns corridors showed in most cases lower R² scores than the utilitarian ones, because of greater variability in the counts. These models are based on data from a single counter rather than a group of them, which may explain the lower scores. Beyond the scope of this paper, it is interesting to note that all utilitarian facilities across the four cities seem to have stagnated or even suffered a decrease in counts during the years 2018-2019, just before the growth in counts caused by the pandemic.

The graph of the utilitarian facilities in the four cities is shown in *Figure 6* and the ones for the Mixed-Utilitarian facilities are shown in *Figure 7*.

City	New York City		Montreal		Ottawa		Vancouver					
	Std.		Std.		Std.		Std.					
Coefficients	Estimate	Error	S.L.	Estimate	Error	S.L.	Estimate	Error	S.L.	Estimate	Error	S.L.
(Intercept)	7.68	0.03	***	6.43	0.05	***	5.53	0.06	***	8.25	0.07	***
site_cat2	-0.25	0.01	***	-0.05	0.01	***	0.44	0.02	***	-	-	-
site_cat3	-0.20	0.01	***	-1.02	0.01	***	-0.35	0.02	***	-	-	-
site_cat5	-	-	-	-0.73	0.01	***	0.00	0.02		-	-	-
site_cat6	-	-	-	-0.20	0.01	***	-	-	-	-	-	-
maximum daily temperature	0.09	0.00	***	0.08	0.00	***	0.08	0.00	***	0.11	0.00	***
sqaured maximum daily temperature	0.00	0.00	***	0.00	0.00	***	0.00	0.00	***	0.00	0.00	***
maximum wind speed	-0.01	0.00	***	-0.01	0.00	***	-0.01	0.00	***	-	-	-
average wind speed	-	-	-	-	-	-	-	-	-	-0.03	0.00	***
precip_cat2	-	-	-	-0.09	0.01	***	-0.07	0.02	***	-	-	-
precip_cat3	-	-	-	-0.51	0.03	***	-0.50	0.04	***	-	-	-
precip_cat4	-	-	-	-0.45	0.12	***	-0.36	0.09	***	-	-	-
precip_cat5	-	-	-	-0.63	0.30	*	-0.83	0.21	***	-	-	-
average daily precipitation	-0.08	0.00	***	-	-	-	-	-	-	-	-	-
average humidity	-	-	-	-	-	-	-	-	-	-0.01	0.00	***
average cloudcover	0.00	0.00	***	-0.01	0.00	***	-0.01	0.00	***	-0.03	0.00	***
total cm of Snow	-0.03	0.00	***	-0.05	0.00	***	-0.02	0.00	***	-	-	-
stringency index	0.00	0.00	**	-0.01	0.00	***	-0.01	0.00	***	-0.18	0.05	***
holiday dummy 1	-0.45	0.02	***	-0.75	0.05	***	-0.40	0.04	***	-0.21	0.09	*
bridge dummy 1	0.12	0.03	***	0.23	0.06	***	-	-	-	0.06	0.03	*
day of the week 2 (Tuesday)	0.04	0.01	***	0.04	0.02	**	-	-	-	0.05	0.03	
day of the week 3 (Wednesday)	0.07	0.01	***	0.04	0.02	**	-	-	-	0.01	0.03	
day of the week 4 (Thursday)	0.03	0.01	**	0.04	0.02	*	-	-	-	-0.06	0.03	*
day of the week 5 (Friday)	-0.01	0.01		-0.05	0.02	**	-	-	-	-0.18	0.03	***
day of the week 6 (Saturday)	-0.20	0.01	***	-0.56	0.02	***	-	-	-	-0.24	0.03	***
day of the week (Sunday)	-0.33	0.01	***	-0.64	0.02	***	-	-	-	-	-	-
weekend_dummy 1	-	-	-	-	-	-	-0.49	0.01	***	-	-	-
season_dummy (spring)	0.10	0.01	***	0.60	0.02	***	0.76	0.02	***	0.35	0.02	***
season_dummy (summer)	0.15	0.01	***	0.71	0.02	***	0.80	0.03	***	0.37	0.03	***
season_dummy (autumn)	0.15	0.01	***	0.89	0.02	***	0.82	0.02	***	0.07	0.02	**
year 2009	-	-	-	0.16	0.05	***	-	-	-	-	-	-
year 2010	-	-	-	0.41	0.04	***	-0.40	0.06	***	-	-	-
year 2011	-	-	-	0.38	0.04	***	-0.09	0.06		-	-	-
year 2012	-	-	-	0.40	0.04	***	-0.01	0.06		-	-	-
year 2013	-0.06	0.04		0.47	0.04	***	-0.06	0.06		-	-	-
year 2014	0.05	0.03		0.46	0.04	***	-0.04	0.06		-	-	-
year 2015	0.14	0.03	***	0.54	0.04	***	0.10	0.06		-0.01	0.03	
year 2016	0.19	0.03	***	0.63	0.04	***	0.03	0.06		0.01	0.03	
year 2017	0.20	0.03	***	0.66	0.04	***	0.18	0.06	**	-0.16	0.03	***
year 2018	0.15	0.03	***	0.54	0.04	***	0.10	0.06		-0.06	0.03	
year 2019	0.16	0.03	***	0.53	0.04	***	0.10	0.06		-0.04	0.03	
year 2020	0.33	0.03	***	0.97	0.05	***	0.60	0.09	***	0.04	0.03	
year 2021	0.37	0.03	***	1.15	0.06	***	0.40	0.09	***	-0.02	0.03	
year 2022	0.36	0.03	***	1.10	0.06	***	0.17	0.08	*	-0.12	0.04	**
2 x log-likelihood [.]	-157726 231		-40893 404		-226507 317		-49342 23					
AIC:	157790		4094	5		-220307.317 226573		49394				
Significance level (S.L.) codes:	·***' 0 001	*' 0 01	0 05	·'∩1	- '' 1	<u>،</u> ،		riahla			-	

Table 2: Coefficients of the Negative Binomial models

Site_cat = variable representing the different counters in the analysis (example MT2, MT3)

Precip_cat: groups of the average daily preciptation. Precip_cat base = no precipitation

Precip_cat2 = precipitation values in the 25% quantile

 $Precip_cat3 = precipitation values between the 25\% the 50\% quantiles$

Precip_cat4 = precipitation values between the 50% and 75% quantiles

Precip_cat5 = precipitation values above the 75% quantiles



Figure 6: Growth of Utilitarian bicycle facilities



Figure 7: Growth in Mixed-Utilitarian bicycle facilities

3.7 Conclusions

This paper investigates the long-term trends across cycling seasons before and after the pandemic using data from counting stations in four cities. Among other things, it was observed that most of the study utilitarian facilities in the four cities suffered a change in the daily and hourly patterns during the first year of the pandemic (2020), which saw a greater mixed and recreational usage of the facilities. This is related to the lockdown and work-from-home measures implemented in the study cities. Despite that, there is no generalized pattern.

Most of the utilitarian facilities have experienced growth in the years before the start of the pandemic; the magnitude of the growth increased in 2020 even after controlling for weather variations across years. This is perhaps related to the shift of public transport users to other modes including cycling during the pandemic. In the year 2022, it seems that the magnitude of the growth as well as the hourly and daily patterns are converging to the pre-pandemic years. The utilitarian cycling corridors in Montreal and New York City grew considerably during the pandemic before stabilizing in the last two years and also returned more quickly to mostly utilitarian patterns after 2020. This could be explained by the fact that both places have the biggest bike infrastructure networks among the studied cities and may have a greater commuter cycling culture. Although the utilitarian facilities in these cities did suffer a sudden change during the pandemic, they do not seem to have been affected by it. They may have even benefited from it.

The mixed-utilitarian bicycle facilities shifted towards more recreational usage during the pandemic. One could attribute this to an increase in recreational riders during the pandemic. However, they do not seem to have experienced any growth in the last years, so it seems that these corridors lost commuters rather than gained recreational users. The changes in the patterns after the pandemic vary between the different cities. While the facility in Montreal shows a tendency towards more utilitarian patterns and a small growth in 2021 and 2022, the growth in the mixed facilities of Ottawa and Vancouver stagnated in the same years and showed a tendency to stay mostly utilitarian or even shift to a more recreational use respectively. In a general sense, these facilities lost commuters during and after the pandemic. Future research analyzing more counting sites of the same type is necessary to confirm this finding.

Ottawa presents a slower recovery to pre-pandemic patterns in most of its sites. Vancouver presented a different situation than the other three cities, where its utilitarian routes have not shown any major growth during the last years, except for 2020. However, the Mixed-Utilitarian routes have grown constantly throughout the years. Also, all the studied sites in this city shifted towards mixed-use during the first year of the pandemic and all three of them do not show any clear sign of changing towards more utilitarian patterns. In general, the studied biking facilities in Vancouver seem to have increased in recreational users. Further studies with data from more counters may be necessary to confirm this finding.

The next steps for this investigation will be to conduct the same methodology with the data from the second half of 2022 and incorporate a greater number of counting sites from other cities in North America. Furthermore, more advanced modelling settings (general additive and dynamic ARIMA regression models) will be tested to account for serial autocorrelation.

3.8 Acknowledgments

The author would like to thank Eco-Counters Inc. for its support with the data access, as well for their help in the computational programming and analysis in this research as well as the MITACS organization for the research financial support.

4. Chapter IV: Bridge

As seen in Chapter III, the cycling demand in big North American cities with significant bicycle networks has grown over the last 10 years. Even with the challenges brought by the COVID-19 pandemic, cycling demand in Montreal grew considerably during those years due to the influx of users shifting from public transit. Although it's not an ideal situation having both modes fighting for the same users, it does show that there is a potential share of people willing to commute using bicycles and that it could be used as a complement to urban public transport. The growth also justifies the future expansion of the cycling network to supply the increasing demand.

Nonetheless, as shown by the literature review, cycling is significantly affected by weather conditions. For example, warmer temperatures increase biking demand and rainy days reduce it. While it is important to control for these variables when evaluating current policies and infrastructure, since favourable weather may hide a decreasing trend in cycling counts (14), city authorities need to acknowledge its effects to ensure that the planned cycling infrastructure continues to meet the needs of its residents in the future. This requires them to understand the growth trends and anticipate the expected demand for cycling facilities in the coming years. Planners should consider the potential impact the changing weather patterns due to climate change will have on future cyclist volumes.

As extreme weather events become more frequent, cities that wish to preserve and grow their cycling commuters will need to adapt the infrastructure to ensure that the bicycle remains a viable mode of transportation throughout the year such as tree-shaded bike lanes, cool pavements, rain-protected facilities, and storm-water drainage. This will require a comprehensive understanding of what weather could be expected in the future to design climate-resilient infrastructure that will protect potential users.

Chapter three showed a methodology that considers the effects of weather on the cycling counts and analyzes the behavioural changes of cyclists in the face of the COVID-19 Pandemic which provides a diagnosis of the present cyclist volumes in North America. The next step is to take this diagnosis into the future to understand

what cities might face in terms of cyclist infrastructure demand and how can they prepare for it.

Unfortunately, the prediction of the forthcoming weather is a tricky task due to all the uncertainties found not only in the interaction between the natural systems but also in possible climate mitigation actions of the nations. Because of this, the expected demand should be estimated across different scenarios and climate models to cover a wide range of feasible outcomes in the middle of all the uncertainty.

Chapter five provides a methodology capable of predicting cyclist counts into any future year of the XXI century by combining historical data from counting sites with the projections given by climate models following different emission scenarios. This framework can calculate a range of expected demands cities must consider when planning the construction of new infrastructure.

5. Chapter V: Effects of Climate Change on Future Yearly Cycling Demand for the City of Montreal

Eduardo Adame Valenzuela

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: eduardo.adamevalenzuela@mail.mcgill.ca

Luis Fernando Miranda Moreno

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: luis.miranda-moreno@mcgill.ca

Van-Thanh-Van Nguyen

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: van.tv.nguyen@mcgill.ca

Chenglong, You

Department of Civil Engineering and Applied Mechanics.

McGill University, Montreal, Canada, H3A 2K6

Email: chenglong.you@mail.mcgill.ca

Data Statement: The cyclist counts data used for this study is the property of Ville de Montreal. Any request for the cycling counts data must be submitted to the corresponding authorities or information portals of each city All the weather measurements data were acquired from the Government of Canada.

Funding Statement: The collaboration between McGill University and Eco-Counters that produced this article was funded by the Mitacs Accelerate program.

5.1 Abstract

In the context of a changing climate, the bike infrastructure requires understanding future needs (demand) under the most probable climate conditions to ensure resilient infrastructure capable of promoting and meeting the expectations of the urban bicycle demand in Canadian cities. The objective of this work is to propose a methodology capable of predicting cycling ridership under alternative climate scenarios by combining models that simulate the relationship between bicycle counts and weather along with climate-model outcomes. Cycling demand projections for 2050 are made using Montreal as a case study. For this purpose, six years of data from nine locations with automatic cyclist counters are used to train a ridership-weather model based on the Extreme Gradient Boost (XGBoost) technique. The predictions of that model are validated by comparing them to known cyclist counts. In parallel, climate predictions for the year 2050 are obtained using the statistically downscaled projected weather conditions from different climate models and scenarios. These climate predictions are inputted into the ridershipweather models to estimate bicycle demand for 2050. From the various scenarios, the results show a change in cyclist counts between -4.77% and 27.22% in cyclist ridership during March, April, and November depending on the scenario considered, with an average increase of 11.21%. During the summer, the change in ridership goes from 9.77% to -11.53%, with an average decrease of 0.88%. Most scenarios predict a slight increase in counts during winter, which can go from 14.55% to -12.02% with an average increase of 1.27%. Depending on the climate scenario, the overall growth in winter, spring, and fall can be offset by the high temperatures during summer. Therefore, results show the need to build infrastructure that considers heat mitigation strategies.

5.2 Introduction

Since the middle of the nineteenth century, the bike has become one of the most recognizable modes of transportation in many cities. The low cost to acquire and maintain a bicycle as well as the several health and mental benefits its usage can provide (26,100–103) makes it appealing for individuals to use. The benefits of the bike are not limited to the user. Higher cycling levels bring many economic benefits to cities (104–106) and reduce air pollution levels in urban areas (107–110). For these reasons, cities have started promoting its use through campaigns, investments in infrastructure, and general policies. An example is Paris, where authorities have developed a plan to transform the city into one of the most bike-friendly environments in the world (111). The "biking renaissance" has also made its way into the North American theatre in the past two decades (6) and Montreal is not an exception. By 2022, the city had built a bike network of 889 km across the whole island (98) and still has plans for future expansion. These efforts resulted in Montreal achieving 18th place in the world according to the Copenhagen Index, making it the only North American city to appear consistent in the index since 2011 (112). However, if authorities wish to continue promoting cycling, they need to plan for the long term.

The problem is that the future hides many challenges, especially in the face of climate change. Therefore, cities need to plan and design more resilient infrastructure. Accordingly, they need to understand better the influence of local weather conditions on cycling demand and the variations (in temperature and precipitation, for instance) that climate change might bring.

The study aims to propose a methodology that helps predict bicycle demand under alternative climate scenarios by combining cycling ridership models and climatemodel outcomes. It consists of three steps; development of ridership-weather models based on historical data, generation of climate projections based on downscaling climate models, and predictions of future bicycle ridership according to the alternative climate scenarios for 2050. The methodology is developed using Montreal as a case study. The results of this paper give insights to the cities about what the future might look like and what features should be considered when planning the bicycle network in the long term.

The paper consists of five sections: 1) a literature review regarding the relationship between the weather variables and cycling demand, estimating cycling demand considering future weather, the expected effects of climate change around the Montreal area, and the use of machine learning models in the prediction of cyclist volumes; 2) the methodology used in this paper; 3) the predictions obtained from the models; and 4) the conclusion of the research, its limitations, possible future research routes, and its implications for future actions regarding infrastructure and transport policy for the city.

5.3 Literature Review

5.3.1 Relationship Between Weather and Cycling

In 1995 Nankervis (11) compared the number of bikes parked inside an Australian university with a daily weather index made up of the maximum temperature, the wind speed and the presence of rain during the day. He found a relationship between the index and bike counts but noted that students are less sensitive to weather variables. Thomas et al. (14) merged daily counts from automatic counters in rural areas of the Netherlands with the daily weather measurements and concluded that the bike counts were mostly affected by the measured temperature as well by precipitation, wind speed and hours of sunshine to a lesser but significant degree. Bean et al. concluded that the hour of the day and precipitation were the most significant variables affecting bike-sharing systems usage in their study across 40 cities (72). For the city of Toronto, Saneinejad et al. concluded that chilly weather, precipitation, and wind speed had a significant impact on active transport modes (113). Wessel found out that even weather forecasts can have a considerable influence on cyclist counts (81). Böcker et al. carried out a literature review on the link between weather and transport and concluded that warm and dry weather increase the use of active transportation modes while rain, snow, clouds, cold weather, and windy days reduce their use. They also noted that bike counts do not have a linear relationship with temperature (114). This idea follows the research of Miranda-Moreno & Nosal (54)

who found a maximum temperature threshold in which, cyclist counts would go down as those variables increase. Nonetheless, this threshold varies by city. The value found by Heaney et al. (82) for the city of New York was similar but not equal to the one found by Miranda-Moreno & Nosal for the city of Montreal. Other cities such as Austin (84) and Seattle (83) also had a threshold value that differed from their eastern North American counterparts. These results reinforce the idea of Goldman & Wessel that cyclists of different cities will have a unique sensitivity to the weather factors depending on the climate of the region or other unseen factors such as the biking culture present (18).

5.3.2 Climate Change Models and Projections Downscaling

As shown, a significant relationship exists between the weather variables (especially temperature and precipitation) and cyclist volumes. While this relationship can be expected to continue, the weather cities will face in the future will not be the same as the present conditions. Climate change is expected to alter the variance and distributions of the weather variables which, as seen in the previous sector, will modify cyclist behaviour. The question that arises is, how will cyclist demand react to the future changing climate and what will this mean for the design of future cyclist infrastructure?

Predicting the weather is a difficult task, especially with a complex system such as the global environment. For this purpose, researchers have developed powerful climate models. These theory-based representations have evolved from conceptual to analogue to mathematical and finally to computational simulations of the atmospheric flow capable of giving projections of possible future weather conditions (115). Nonetheless trying to predict the weather in the years to come is more difficult since the influence of human activity has been proven to alter the climate and induce warming of global temperatures (116). For this reason, various models have been run considering different scenarios of human activity to cover a wide range of possible future climates. Although these models represent powerful tools, their output cannot be used at a local level due to its coarse spatial resolution of about 200 km (117). That cell size is unable to capture the variations at a lower scale, which is needed when studying the future weather in smaller spaces such as urban areas. Downscaling allows establishing links between the climate model outputs with the surface temperature in an area. This can be done in two ways: The first strategy is called dynamical downscaling (DD) which follows the rules of conservation of mass-energy and momentum by creating a smaller model spanning across the desired area and using the outputs of the larger models as boundary conditions. Although they allow for more complete physics, they require high amounts of computational power. Statistical Downscaling is a second, less computationally intense strategy that provides faster results by using statistical regression techniques to model the relationships between the model projections and the surface temperature (118). Statistical downscaling techniques can be divided into three groups: Regression models that linearly represent the relationship directly between the outputs of the climate model with the conditions at the point for each weather factor, Stochastic Weather Generators that simulate weather data based on the characteristics of the variable, and weather typing that groups days into weather condition types and the bigger climate models are used to determine the frequency and variance of these types (119).

The models and downscaling methods have been used in the literature to provide insight into what the weather might look like in the future for a city like Montreal. In a general review of the effects of climate change on the weather conditions of the United States, Huber and Gulledge (120) concluded that there is a statistical trend towards more frequent and intense extreme weather events in the future and an increase in the risk of heatwaves over time. The authors also mentioned that cold events will become less frequent but will not disappear and noticed that the models tend to under-predict maximum temperatures. Berardi and Jafarpur (15) statistically downscaled a General Circulation Model (GMC) to obtain future weather values for the city of Toronto. They projected an increase in the mean temperature of 0.8 °C and a decrease in the humidity values and concluded that the combination of multiple

GMCs along with Regional Climate Models (RCM) had the best prediction accuracy of the climate in the region. Wang et al. (121) developed high-resolution climate projections for the province of Ontario using the HadCM3 model and the Providing Regional Climate for Impact Studies (PRECIS) regional modelling system to downscale the weather projections. Their results showed a province-wide increase of the average temperature of 2.6 to 2.7 °C for 2030, 4 to 4.7°C in 2050, and 5.9 to 7.4 in 2080. They also projected a slight increase in the annual sum of rainfall, a greater number of precipitation events during the winter and spring months, as well as a province-wide increase in rainfall intensity. Cheng et al. (122) used a synoptic weather typing method's capability to analyze the impact of climate change on extreme events in the south-central region of Canada. They estimated an increase in heat-related mortality, freezing rain as well as the frequency and intensity of future daily extreme weather events. For Montreal, Gaitán (123) downscaled the CGCM3.1 and CRCM4.2 models and found an increase of 7°C in the maximum temperature for a 30-year return period. Dickau et al. (124) used a MarkSim Weather Generator to obtain future temperature projections and estimated a reduction in the number of days of the outdoor skating season due to the projected increases in temperature and the shortening of the freezing season.

5.3.3 Climate and Cyclists

These studies confirm a growing trend in the number of warm days and a shortening of the typical winter season which suggests an annual increase of cyclist counts with more equally distributed volumes across the seasons. This might cause an extension at both ends of the biking season in Montreal, which has historically taken place from mid-April to mid-November. Nonetheless, the models also predict an increase in total precipitation and frequency throughout the year as well as an increase in the number of heat wave events during the summer which could counteract the positive effects of a warmer shoulder season (spring and fall) and a milder winter.

Böcker et al. (114) studied the potential effects of climate change in modal choice and distance travelled by mode for the year 2050 in the region of Randstad in the Netherlands using extreme observed weather values to represent the expected

weather for 2050. Their results showed an increase in the modal share of active transport during the winter and a loss during the summer caused by high temperatures and greater precipitation volumes. Wadud calculated the effects of climate change on cycling levels in London for 2041 with negative binomial models and data from the HadCM3 general climate model. He predicted an overall marginal annual increase in counts due to the counteracting effects of temperature and precipitation with cyclist volumes increasing during the summer and winter and decreasing in the spring and autumn (125). One major difference with the work of Böcker et al. is the use of data from future projections made by a climate model and from automatic bike counters instead of survey data, which is the standard in the following papers. Heany et al. (82) estimated the future number of trips and their length for the Bike Sharing System (BSS) of New York City using a general additive model. When inputting future weather conditions into the model, the results showed an annual increase in the number of trips and their length. On a seasonal level, they predicted a decrease in the summer counts due to extremely hot days, which was offset by the gains in spring, autumn, and especially winter. Similarly, Galich et al. (126) predicted the bicycle counts in Berlin for 2050 while adjusting for climate change with four different models: a statistical one and 3 machine learning based. The authors concluded that the city would experience a small annual increase in bike counts between 1 and 4% compared to 2020, with greater gains being made during the winter season. Additionally, they concluded that the machine learning models had a greater predictability accuracy. Sharafi (127) predicted trip counts for the BSS in Montreal using statistical regression models and future weather projections derived from a Global Environmental Multiscale model. She concluded that future weather would increase the demand for the BSS in April, September, and October. She did not study the winter months since the BSS closes during that season and did not consider non-BSS cyclists. Chan and Wichman (128) quantified the monetary impacts of climate change on leisure cycling in 16 cities across North America with the use of a regression model. On the seasonal scale, they predicted greater cyclist volumes throughout the year with a marginal change in summer, greater in the shoulder months and the greatest during the winter. Nonetheless, it is

important to notice that their analysis across multiple cities and climate zones may have not accurately projected the responses of the citizens from each city to the changing climate since it is not the same to have warmer days in a cold city than in one with already extremely hot days. As a comparison, Mathisen et al. (129) estimated the bike usage in the city of Bodo, located in the Arctic region of Norway for low, middle, and high emissions climate projections. The authors found a small increase in bike usage in all the scenarios. On the other hand, Lanza et al. (84) carried out similar research to the previous papers on an urban trail located in the city of Austin, United States, which is in a humid sub-tropical area. The authors imputed the expected climate condition obtained from downscaled climate model predictions for the 4.5 and 8.5 emission pathways of the IPCC. They estimated a decrease in urban trail usage due to the loss in counts due to the extremely hot temperatures of the summer offsetting the gains during warmer days in the winter.

The revised papers seem to agree that in cities with continental or colder climates, like Montreal, the annual bike usage will slightly increase due to climate change with the greatest gains during the winter. Nonetheless, the impact of climate change on counts from one city cannot be assumed to be true for another since the variation in the climate and climate resilience of the citizens will be different. The magnitude of the effect also varies throughout the year, which shows the importance of carrying out separate analyses for each season, as noted by Böcker et al. (114). Additionally, it is important to consider the observation of Mathisen et al. (129) that their model had a bias toward overestimating low counts, which are more likely to occur during the winter. If all the models had a similar problem and the predicted values were compared to the observed values rather than to the fitted values, there would be an overestimation of the impact during the low-counts seasons, which could explain the great increase during the winter season reported by the articles. The literature proves that the weather conditions in the region and its variations due to climate change can influence the prediction of future annual counts. This information confirms that cyclist counts are likely to change in the future and supports the need to carry out this type of analysis for Montreal.

5.3.4 Machine Learning Methods and Cyclist Predictions

Researchers have started to explore the capabilities of newer models based on machine learning methods to predict cyclist demand. A study used bicycle renting data from Washington D.C. and compared the prediction capabilities of linear regression (LR), random forest (RF), and extreme gradient boosting (XGBoost) models, with the latter one outperforming the others (130). Another study also compared different models trained on data from Brussels, Belgium along with measured weather factors. The authors mentioned that models' prediction capabilities increased after adding the weather variables with the Fully Connected Neural Network (FNN) performed the best, followed by the XGBoost model (131). A different study paired a database of bicycle trips in Washington D.C. with weather variables and found that the XGBoost model had the best RMSLE compared to LM, RF, Support Vector Regression (SVR), Adaboost, and bagging models (132). Cui et al. developed an XGBoost prediction model of cyclist demand at the exits of subway stations in Beijing, China, which performed better than other statistical models. They also mentioned that special weather events had a noticeable impact on cyclist counts, and concluded that adding these factors to the model will increase its prediction capabilities (133). In Seoul, South Korea, researchers compared different models trained on hourly bike data while controlling for weather effects and concluded that gradient boosting, followed by XGBoost, performed better than LM, SVR, and boosted trees. They also found that the temperature and hour of the day attributes had the greatest prediction capability (134). A study using bike rental data from Thessaloniki, Greece, included the measured weather in the analysis and compared different machine learning models (Random Forest, Gradient Boosting, Extreme Gradient Boosting and Neural Networks). They concluded that gradient boosting and XGBoost had the best prediction capabilities (135).

5.4 Methodology

5.4.1 Definition of the Periods

The study defines three distinct periods. The first one is *The Training Period*, (from 2015 to 2018) used to fit the models. The second is *The Validation Period* (2019),

used to evaluate the model and to compare the changes with the year 2050. These two periods compose *Historical Data* (from the 1st of January 2015 to the 31st of December 2019) whose years were selected to maximize the amount of data. The years 2020 and 2021 were excluded due to the lockdown measures of the COVID-19 pandemic. The third one is *The Prediction Period*, which goes from the 1st of January 2050 to the 31st of December 2050 in which future cyclist volumes were estimated.

5.4.2 Selection of the Sites and Cleaning the Data

The data information from the 53 automated bicycle counters owned by the City of Montreal and fabricated by Eco-Counters were downloaded. Each site was classified according to its daily and weekly profiles with the same methodology as a previous study (136). The data were cleaned following two processes. First, all zero counts observed outside of the winter period were considered non-representative and replaced with an NA value. Secondly, for each counter, the observations were compared to an upper and lower threshold that was calculated per season as shown in *Equations 1* and 2:

Upper threshold = Q_{75} + 1.5 * IQR [eq. 1] Lower threshold = Q_{25} - 1.5 * IQR [eq. 2]

Where Q₇₅ is the 75th quantile of the counts, Q₂₅ is the 25th quantile, and IQR is the Inter-Quantile Range which is equal to the difference between Q75 and Q25. This method was chosen because it does not assume any distribution of the data. Any daily counts outside their corresponding seasonal thresholds were replaced by an NA value. Afterwards, a linear interpolation was made in cases where the number of consecutive days with NA values was less or equal to three. All other NA values were eliminated leading to a deletion of 4.69% of the data. Two rules were followed to select the final list of counters to be used in the study:

- 1) The counter should have more than 90% of the data from the year 2015 to 2019.
- 2) The counter should be used mostly for utilitarian purposes.

In the end, the data from nine counters were selected. The counter with the lowest amount of non-missing data accounted for 90.7% of the counts of the historical period and the highest for 99.54% with an average of 96.3% and a total of 18,986 observations, 15,962 in the Training Period and 3,024 in the Validation Period.

The cleaned historic counts were paired with the measured weather variables registered by the station located at the Montreal-Trudeau International Airport. Afterwards, the data sets were divided into *Training Period* and *Validation Period* based on the periods defined above. To account for the seasonality, variables such as the week, month of the year, holiday, and bridge day (if the day was a Friday that followed a holiday on Thursday or a Monday followed by a holiday on Tuesday) were added, as well as the longitude and latitude coordinates of each site to control for spatial correlations.

5.4.3 Ridership-Weather Model

Based on the literature, machine learning methods have shown a greater predictability capacity compared to statistical ones. However, the fitting process is more complex, and the results cannot be easily translated to establish the elasticities of the dependent variable with the independent variable. Furthermore, they have a higher risk of overfitting the sample data (137), which means that the input of different data would not alter by much the result given by the model. The XGBoost algorithm already implements strategies such as the regulating objective, scaling of the weights for each step, and the subsampling of independent variables (or features) for each step of the model (138). The last two can be tuned as hyperparameters to optimize the results and reduce overfitting. The hyperparameters for this study were estimated using a randomized grid search by first selecting 100 random hyperparameter combinations. Each combination was used within a stratified five k-fold cross-validation (xgb.cv function from the xgboost package for R). In other words, the training data were split into 5 groups with more or less the same distribution. Afterwards, a model was trained with 4 of those 5 groups considering the combination to set the model's hyperparameters and established a *train RSME* as a measure of how well the model fitted the training data.

Then the newly trained model predicted the counts using the independent variable values stored in the fifth group of data and was compared to the observed counts to set the test RSME. This process was repeated four times more so each group of data would be excluded at least once from the training process and used as testing data. A final test-train ratio was calculated by dividing the Average Test RMSE by the Average Train RMSE of the five folds. The same process was carried out with all the other 99 combinations. Then the 5% combinations with the worst (highest) Average Train RMSE were eliminated due to underfitting. After, the 25% with the best ratio (those closer to 1) were kept to avoid overfitting. Finally, the combination with the best Average Test RSME of that group was selected. A new random set of 99 combinations was created with values close to the ones selected in the previous round and joined together to get a total of 100 combinations. The process was repeated two times to get a final selection after the third round. Although the random grid search may select a local best combination instead of the absolute best, it is considerably less computationally expensive and still yields models with good results.

The final hyper-parameters selected obtained an *Average Train RMSE* of 407.18 and an *Average Test RMSE* of 417.47. These were: maximum levels per tree (*max_depth*) equal to 1, the percentage of features used per tree (*colsample_bytree*) equal to 0.6, the percentage of observations used to train each tree (subsample) equal to 0.85, minimum weight per node (*min_child_weight*) of 3, a learning rate (eta) of 0.6, a gamma value of 4, and a total number of iterations (*n_rounds*) equal to 850. The Training Data were separated into four different seasons: winter, spring, summer, and autumn which were used to fit four different models.

Afterwards, the four seasonal models predicted the 2019 counts using the independent variables of the *Validation Period*. The results obtained a Mean Average Percentage Error (MAPE) of 42%, a Median Average Percentage Error (MDAPE) of 19% and a coefficient of determination (R²) of 0.93 when compared to the observed counts for the same period.

5.4.4 Selection of the Climate Models and Downscaling

The CanESM2 (Canada) and the HadCM3 (U.K.) Global Climate Models were selected because they provide values for different emissions scenarios and have been validated for a longer period than newer models and by multiple researchers. The projections made by the two models consider different emission scenarios.

Model	Scenario	Description							
	A2	Storyline of a heterogeneous world with countries working on							
		regional and local solutions with a policy focused on econor							
HadCM3		growth (139).							
	B2	Storyline of a heterogeneous world with countries working on							
		regional and local solutions with a policy focused on							
		environmentally sustainable growth and slower population							
		growth (139).							
	RCP 2.6	Strong emission mitigation strategies were implemented with an							
		average radiative forcing level of 2.6 W/m ² by 2100(46).							
CanESM2	RCP 4.6	Intermediate emission mitigation strategies implemented with an							
		average radiative forcing level of 4.5 W/m ² by 2100 (46).							
	RCP 8.5	Weak emission mitigation strategies were implemented with an							
		average radiative forcing level of 8.5 W/m ² by 2100 (46).							

While one could classify the models into low emissions (CanESM2 RCP 2.6 and HadCM3 B2) and high emissions (CanESM2 RCP 4.5, RCP 8.5 as well as HadCM3 A2) it would be a mistake to assume them equal since they are based on different assumptions, so all comparisons must be made between scenarios from the same model.

The projections of all the scenarios for both models are available in resolution 50 km² grids. For the study, the projections were needed on a smaller scale to cover the island of Montreal. The climate values were statistically downscaled using SDSM software. This required that the climate projections along with their historic validation values be compared and regressed to the historic measured weather in Montreal.

The historic weather data were obtained from the Dorval Weather Station, located in the Montreal–Trudeau International Airport from 1941 to 2019. Weather data from the same weather station was used to train and validate the cycling demand prediction model. Five datasets with the downscaled weather measurements for each scenario from both climate models were created for the *Future Data* by paring the projected daily maximum temperature and daily precipitation accumulation for the year 2050 were matched to the additional regressors for 2019 data by day of the year.

5.4.5 Prediction and Comparison

The future data sets were divided into four subsets, one for each season and input into the validated models to predict the cycling counts for every day of the year 2050 with the four future datasets. The predictions were compared against three sets of 2019 counts. First, the observed as registered by the automatic counters which are the true values. Second, against the counts predicted by the validated models using the measured weather to eliminate the cyclist prediction models' error while still evaluating close to the real situation. Third, against the predictions made by the cyclist demand models using the weather projections for 2019 from the same climate model and scenario to correct not only for any error in the cyclist models but also for the bias present in the climate model. The percent changes were calculated at the annual and monthly levels.

5.5 Results

5.5.1 Climate Projections Comparison

Compared to the measured weather in 2019, we can see that all the models do a good job capturing the seasonality of the maximum daily temperature, as shown in *Figure 8* (validation refers to the projected weather for and prediction of the projected weather for 2050).



Figure 8: Average max daily temperatures per month.

Regarding the daily accumulated precipitation, the values for the 25th, 50th, and 75th percentiles are shown in *Table 3*.

Source	Period	Q 25th	Q 50th	Q 75th
Measured	2019	0.00	0.00	2.80
CanESM2_rcp26	2019	0.00	0.06	4.06
CanESM2_rcp45	2019	0.00	0.00	2.91
CanESM2_rcp85	2019	0.00	0.00	1.56
HadCAM3_B2	2019	0.00	0.00	3.30
HadCAM3_A2	2019	0.00	0.00	2.33
CanESM2_rcp26	2050	0.00	0.00	2.83
CanESM2_rcp45	2050	0.00	0.00	3.30
CanESM2_rcp85	2050	0.00	0.31	3.74
HadCAM3_B2	2050	0.00	0.00	2.43
HadCAM3_A2	2050	0.00	0.00	4.00

Table 3: Percentiles of the daily precipitation distribution.

5.5.2 Validation of the Daily Cyclist Counts Model

The information from those six datasets (measured weather in 2019 and the downscaled projections from each of the five climate models for the same year) was used as input to generate six sets of predictions for the 2019 daily counts, with all other variables staying the same. *Figure 9* shows the time series of the predictions made by the model using the 2019 weather data.



Figure 9: Model validation time series and metrics.

2019 Weather source	R2	MAPE	MDAPE
Measured	0.93	0.42	0.19
Projections CanESM2_rcp26	0.79	0.66	0.31
Projections CanESM2_rcp45	0.79	0.67	0.29
Projections CanESM2_rcp85	0.83	0.58	0.26

Table 4: Validations' R squared values.

Projections HadCAM3_A2	0.82	0.62	0.28
Projections HadCAM3_B2	0.79	0.67	0.31

From the values presented in *Table 4*, we can see that the predictions given by the model using the measured weather have a high coefficient of determination (R2) to the observed counts and follow the seasonality of the cyclist demand throughout the year in the different sites. As expected, the different weather projections produced by the models yielded different results; however, all R2 values remained equal to or above 0.78 and the highest median percentage error (MDAPE) was 30 %. *Figure 10* shows the boxplots of the daily counts observed and predicted in all the sites using the six weather data sets for the year 2019.



Figure 10: Boxplots for the 2019 daily counts.

5.5.3 Predictions for 2050.

The five different 2050 weather projections from the downscaled climate models were input into the 2019 regressors dataset. This means that the models predicted the counts as if the year 2019 had the weather of 2050, similar to the method followed by Galich et al. (126). *Figure 11* shows the boxplots of the daily counts observed in 2019 and those predicted for the year 2050 in all the sites.



Figure 11: Boxplots for the 2050 daily counts.

In *Figure 11* the boxplots show a small decrease between 3 to 4% of the 75th quartile, an increase between 15% and 19% for the 25th quantile and an increase between 20 to 23% of the median for the CanESM2 scenarios compared to the 2019 observed counts. In the HadCM3 scenarios, one can see that the quantiles for the A2 scenario changed by 11%, 25% and -1% (25th, 50th, and 75th quantiles respectively) while for the B2 scenario, the count's quantiles changed by 6%, 11%, and -3% (25th, 50th, and 75th quantiles respectively) for the year 2050.



5.5.4 Comparison Between 2019 and 2050

Figure 12: Average daily counts predicted for 2019 and 2050 for the three CanESM2 scenarios.

Figure 12 shows the time-series bar graphs with the monthly average daily counts for the three CanESM2 scenarios. One can see that the weather projected in the low emission scenarios causes higher counts in the future for May and June, while the other two scenarios show lower expected counts in June, July and August compared to 2019 while having higher counts in November.



Figure 13: Percent change between 2019 and 2050 counts for the CanESM2 scenarios.

Figure 13 shows the percent change per month between the 2019 counts and those predicted with the projected 2050 weather. In a general sense we can see that for the first scenario, the percent change around the summer stays closer with positive changes in May and June, while the other scenarios show a more consistent growth in February, March, and May.



Figure 14: Average daily counts predicted for 2019 and 2050 for the two HadCAM3 scenarios.



Figure 15: Percentage change between 2019 and 2050 counts for the HadCAM3 B2 (top) and A2 (bottom) scenarios.

As shown in *Figures 14* and *15*, the high emission scenario (A2) causes greater cyclist volume gains in the shoulder months of March, May, and November, while

the low emission (B2) scenario has smaller negative changes during the summer. Nonetheless, both scenarios show a reduction in counts during the summer months, especially in July and August.

Model	Observed	2019	2019	2050	Compared	Compared	Compared
	2019	Predictions	Predictions	Prediction	to	to	to
		with	with	with	Observed	Measured	Projected
		Measured	Projected	Projected			
		weather	Weather	Weather			
CanESM2_rcp26	1506	1472	1479	1528	1.46%	3.81%	3.30%
CanESM2_rcp45	1506	1472	1551	1492	-0.94%	1.36%	-3.79%
CanESM2_rcp85	1506	1472	1570	1505	-0.07%	2.24%	-4.13%
HadCM3_B2	1506	1472	1534	1526	1.31%	3.66%	-0.51%
HadCM3_A2	1506	1472	1536	1539	2.21%	4.57%	0.21%

Table 5: Observed and predicted counts for each model.

Table 5 shows in numbers the average daily counts for 2019 observed at the counting sites as well as those predicted by the Cyclist Counts Model using the measured weather data and the projected data from the same climate model and scenario. The fifth column shows the predicted counts using the 2050 projected weather. The final three columns show the percent changes when comparing the counts in the fifth column with those of the second, third and fourth columns respectively. One should notice that, for the CanESM2 climate model, only the RCO 2.6 scenarios expect a general increase of count ranging from 1.45% to 3.30% while the other higher emission scenarios show, in general, a reduction of count into the future. For the HadCAM3 models, the "higher emission" scenario causes greater growth compared to the B2 one. This could be explained that, in this climate model, temperatures during the summer are not expected to increase as much as in the CanESM2 model, so the gains in the shoulder month are enough to offset the losses in summer. However, it is a very marginal gain. In general, rising temperatures will not necessarily lead to an annual growth of cyclist numbers in the future at these counting sites since the summer might become too warm to bike.


Figure 16: Daily average cyclist volumes per month for each climate model.

Figure 16 also shows the predictions calculated for the year 2050 using the weather projections from the different models and scenarios. During the summer months of June and July, the low-emission scenarios (represented in the lighter colours) predict higher average daily counts than their high-emission counterparts, while the inverse is generally true for the shoulder months. Another similarity is the average daily counts during the winter months through the different models do not represent the greatest gains which differs from the findings reported in other studies (82,126,128). However, none of those studies were carried out in the same region as Montreal, so the winter experience in those cities may not be as cold as those experienced in Québec. On the other hand, the study's finding of the reduction in counts during the summer months does align with the findings of similar studies (82,84,114,128) and the finding of the greatest increases occurring during the shoulder months also agrees with the conclusion drawn by Sharafi for Montreal (127).

5.6 Conclusion

The results suggest that, when considering the projections of the CanESM2 model, the weather in the high-emissions scenarios increases the counts of cyclists in March, April, and November, while the lower-emission one shows a higher number of cyclists in June and July. Given that the cycling demand is much higher during the summer months in Montreal, the annual ridership is expected to decrease between 0.65% and 1.12% compared to the reference period in the higher emission scenarios.

The predictions under the HadCAM3 models show an expected annual cycling increase of 2.33% for the A2 scenario and 1.49% for the B2 scenario. Both scenarios estimate an increase in counts during April, May, and June going from 6.71% to 25.4%, and a mostly negative change in August and September between 2.59% and -4.33%. Only the A2 scenario estimates a growth during winter between 0.38% to 13.82%, which can be explained by the higher temperature expected for that scenario.

For both climate models, the greatest increase in cycling volumes occurs during the shoulder months of the cycling season (March, April, and November). However, in the CanESM2 high emissions scenarios these gains are not enough to offset the percentage loss in ridership during the summer. This loss of counts during the summer months in the high emissions scenarios can be explained by a considerable number of days that are expected with maximum temperatures greater than the threshold found by Miranda-Moreno and Nosal (54). Additionally, heavier precipitation events will also decrease counts even if the days are warmer.

These results suggest that cities can expect a stretch of the cycling season in the shoulder months. Nonetheless, the increased temperatures in the summer seem to decrease counts which is explained by the significant relationship between cyclist volumes and Universal Thermal Comfort Index (140). Accordingly, if cities like Montreal wish to maintain a higher number of cyclists throughout the year, cities must adapt the infrastructure to offset the effect of expected higher temperatures during the summer and higher frequency of heavier rain events. One possible

improvement is the use of green cover, which could eliminate between half and onethird of the extra-urban heat island effect for the year 2050 (141). Other strategies are to change the asphalt pavement for lower heat-absorbing infrastructure such as interlocked bricks (142) or other cool pavement strategies capable of mitigating urban heat discomfort (143). The use of permeable and high albedo surfaces can reduce urban temperatures. Multiple strategies can be incorporated beyond the bike lanes such as white cool roofs, water bodies, and green pavements (144) to reduce the land coverage of artificial heat-absorbing materials. It should be noted that, in any scenario, the change in annual cyclists is marginal with a maximum of 4.13%, which is smaller than the accuracy of the models. The results show that the estimated impact of climate change on cycling can be smaller than the prediction error of the model.

While this study has tried to provide results as accurately as possible, it is important to note the current limitation of the climate models to accurately predict long-term climate changes, especially when considering precipitation. The results shown here are based on the available data and models and best estimates up to date. However, caution must always be taken when dealing with weather estimates. This study does not consider any inherent increases in bicycle ridership that might result from other urban design and land use policies, investments in bicycle infrastructure that could accelerate bicycle urban mobility, actions to improve road safety of active modes, etc. Also, this study assumes that the relationship between the cyclist number and the weather will remain the same from 2019 up to 2050. Yet, this may not be the case since the citizens may acclimatize to the future weather, an idea discussed by Goldman & Wessel (18).

Other lines of research could focus on determining the cold "threshold", how warm a day must be before one can observe a substantial increase in the counts. Future research will need to be carried out as newer and better versions of the climate models and machine learning technique models are made available to the research community.

5.7 Acknowledgements

The author would like to thank Eco-Counters Inc. for their support with the data access, and data analysis insights for this research as well as MITACS organization for the financial support.

6. Chapter VI: Discussion of the Findings

This thesis presents a framework to investigate the historical bicycle patterns in North American cities and the impact of weather conditions on urban cycling to predict the potential effects of climate change. For this purpose, the methodological contributions of this work are divided into two components: i) improvement of the understanding of bicycle trends and the effect of weather in the years before and after the COVID-19 pandemic, and ii) the prediction of bicycle demand changes under different climate scenarios for the year 2050. Using data from several counting stations in the Cities of Montreal, New York, Ottawa, and Vancouver, along with weather measurements from nearby stations, this method can determine the accurate trends in cyclist demand regardless of the effect of weather. This work requires the integration of different sources of data, modelling techniques and downscaling climate models.

In the first part of this thesis, the proposed methodology proved cyclist volumes in the counting stations analyzed in the study are increasing in Montreal regardless of the weather conditions and lockdown measures, with significant growth in 2021 compared to 2009 volumes when controlling for weather factors. The demand increased noticeably in 2020 at the height of the COVID-19 pandemic, which is explained by a modal shift of public transport users towards an individual, but still cheap, transportation alternative. This phenomenon shows a potential group willing to incorporate cycling into their commute. Another finding was that, even at the height of the pandemic in 2020 when work-from-home policies were in place, most of the analyzed cycling facilities retained a mainly utilitarian use and those who shifted towards recreational purposes showed signs of reverting to pre-pandemic patterns in 2021 which confirms a strong commuting culture within the city. These two points proved the existence of a strong and growing commuter cyclist group in Montreal that will require the provision of safe and reliable cycling infrastructure in the coming years. If the city wishes to continue serving this population, the facilities will need to be expanded to accommodate a constantly growing population in the next few years. However, climate change will introduce variations in the weather that could alter the annual bike-use behaviours in the medium and long term. Any longlasting infrastructure will need to be designed with these changes in mind to expand the project's useful lifetime.

It should be noted that the classification method used, based on the temporal distribution of cyclists has some limitations. First, it is biased towards jobs with more traditional 9 to 5 schedules which does not encapsulate all labour positions. This could leave trips made by population groups that work on a different schedule outside of the analysis, which could lead to ignoring the trip behaviour of a sector of the population. Additionally, the pandemic could have shifted the traditional working hours in the last few years, as more flexible schedules are adopted thanks to work-from-home or hybrid work schemes. This would require other sources of data such as surveys to validate de classification of the counting sites and the behavioiral patterns of the employed population. Nonetheless, due to the available data, it was deemed the best classification method available for most of the trips.

Either way, one can assume that the classification shows which used is most likely given by the cyclist based on traditional work schedules. For the Vancouver results, insights from the locations have noted the existence of a bypass used by commuters to avoid leisure traffic at the VN1 and VN2 counter sites, which would explain why they showed the biggest recreational use compared to all the other sites in the study. Furthermore, both sides are close together, which could have influenced the analysis. These observations support the need to repeat the methodology in the city with different sites and more precise knowledge of the area to reach better conclusions.

In the second part of this thesis, the proposed methodology helps describe the future changes in the cycling demand considering the expected variations in the weather factors under critical climate change scenarios. The results from the different scenarios show a major percentage increase in counts during the shoulder month which suggests that, in the future, the biking season in the city will start earlier and finish later in the year. Although it would be logical to assume that the winter season would see the greatest increment in counts due to rising daily temperatures, the analysis shows a smaller percentage change compared to the gains observed in

November and April. One possible explanation is that, even if the days' maximum temperatures are warmer compared to the present, most days will still be too cold to bike, and precipitation will play a bigger role in those freezing days. The increase in temperature during the spring and autumn might bring most days into an optimal cycling season into the "Goldilocks" range. Nonetheless, the results also show that, with higher global average temperatures, counts during the summer months are predicted to decrease. As discussed in the literature review, the relationship between the temperature and cycling number is not linear and, after reaching a threshold, counts will start to decline as the temperature rises. The increase in daily maximum temperature would cause greater heat discomfort to the potential cyclists, pushing these days out of the ideal weather conditions for cycling. The combination of these factors could lead to a reduction in annual counts in the future since the percent gain of cyclists in the shoulder months is not enough to offset the losses during the summer, which currently has the highest number of daily counts.

7. Chapter VII: Conclusion and Summary

The thesis highlights a bicycle commuting segment of the Montreal cyclist population that has consistently grown during the last decade. The growing trends reinforce the positive impacts of the current strategies to expand the cycling network to promote and accommodate additional cycling demand in the coming years.

The results of this research can be valuable to authorities in their infrastructure planning endeavours. By considering the trends observed in recent years and projecting them into the future, cities can better anticipate the demand changes and needs of infrastructure development. Furthermore, accounting for the changing weather patterns in these projections increases the accuracy of the analysis and ensures that infrastructure plans are well-suited to meet future challenges in a changing climate context.

The methodologies used in this thesis can serve as a guideline for analyzing current growth trends in cycling ridership while accounting for weather conditions in other cities. By incorporating these methodologies, authorities can gain a deeper understanding of how weather and other factors influence cyclist ridership in the present and make more accurate predictions about future demand.

Regarding the future, different scenarios suggest an expansion of the biking season beyond the current time frame, currently from April 1st to November 15th (145). City authorities should keep promoting the maintenance of cycling infrastructure across seasons to ensure they remain open and in optimal condition for an extended period of the year to accommodate for a possible increase in demand. More importantly, to promote cycling as a viable transportation mode throughout the year, planners will need to incorporate strategies into the existing and current facilities, such as heat-mitigating strategies during summer. As temperatures rise due to climate change and natural weather cycles, it is essential to consider designs that enhance thermal comfort for cyclists. Approaches like green shading and cool pavements can help mitigate the heat, making cycling more comfortable even during those hotter months. These strategies would retain a greater volume of cyclists during summer and lead to an overall gain in annual cyclist counts.

Overall, the research presented in this thesis expects to provide valuable insights that can promote cycling as a key element in sustainable urban mobility solutions. By effectively designing infrastructure to meet the expected demand and considering the impacts of changing weather patterns, authorities can continue to support and encourage the use of bicycles as a viable mode of transportation in Montreal. Greater adoption of this mode year-round will bring multiple benefits to the city, like the reduction in healthcare costs, lower inner-city emissions and congestion levels, as well as smaller land consumption dedicated to the transportation of people.

Future lines of research would include the evaluation of the growth trends in the postpandemic years of 2022 to 2024 to confirm if the observed trends remained as expected or turned out to just be a short-phase phenomenon. The same methodology of the first paper could be expanded to include more counting sites in Vancouver and New York as more of the data are cleaned and reconstructed to confirm the observed trends in those cities. In the same way, the methodology could be used to carry out the analysis in different cities around the world to estimate the growth trends while controlling for the effects of the weather. Regarding prediction into the future, the methodology of the second objective could be repeated for different future years to estimate an annual time series and variations in counts to understand the temporal trends expected due to climate change under different scenarios. Additionally, the research should be repeated whenever more accurate climate projections are made available and as the models' prediction capabilities increase with the adoption of new technology. The proposed methodologies could also be automated and integrated into a decision-support toolbox that could make the data, models, and outcomes easily accessible to decision-makers.

8. Bibliography

- 1. LeGates RT, Stout F, editors. The City Reader. 7th ed. London: Routledge; 2020. 808 p.
- Angel S, Parent J, Civco DL, Blei A, Potere D. The dimensions of global urban expansion: Estimates and projections for all countries, 2000–2050. Prog Plan. 2011 Feb 1;75(2):53–107.
- 3. Litman T. Understanding Smart Growth Savings. Victoria, B.C., Canada: Victoria Transport Policy Institute; 2023 Mar.
- 4. Frank LD, Engelke PO. The Built Environment and Human Activity Patterns: Exploring the Impacts of Urban Form on Public Health. J Plan Lit. 2001 Nov 1;16(2):202–18.
- 5. Beauregard RA. When America Became Suburban [Internet]. Minneapolis: University of Minnesota Press; 2006 [cited 2023 Apr 4]. Available from: https://muse.jhu.edu/pub/23/monograph/book/27664
- 6. Pucher J, Buehler R, Seinen M. Bicycling renaissance in North America? An update and re-appraisal of cycling trends and policies. Transp Res Part Policy Pract. 2011 Jul;45(6):451–75.
- 7. Blondiau T, van Zeebroeck B, Haubold H. Economic Benefits of Increased Cycling. Transp Res Procedia. 2016 Jan 1;14:2306–13.
- 8. Mueller N, Rojas-Rueda D, Salmon M, Martinez D, Ambros A, Brand C, et al. Health impact assessment of cycling network expansions in European cities. Prev Med. 2018 Apr 1;109:62–70.
- 9. Deenihan G, Caulfield B. Estimating the health economic benefits of cycling. J Transp Health. 2014 Jun 1;1(2):141–9.
- 10. Pucher J, Buehler R. Cycling towards a more sustainable transport future. Transp Rev. 2017 Nov 2;37(6):689–94.
- 11. Nankervis M. The effect of weather and climate on bicycle commuting. 1999;15.
- Meng M, Zhang J, Wong YD, Au PH. Effect of weather conditions and weather forecast on cycling travel behavior in Singapore. Int J Sustain Transp. 2016 Oct 20;10(9):773–80.

- 13. Nosal T, Miranda-Moreno LF. The effect of weather on the use of North American bicycle facilities: A multi-city analysis using automatic counts. Transp Res Part Policy Pract. 2014 Aug;66:213–25.
- 14. Thomas T, Jaarsma R, Tutert B. Exploring temporal fluctuations of daily cycling demand on Dutch cycle paths: the influence of weather on cycling. Transportation. 2013 Jan;40(1):1–22.
- Berardi U, Jafarpur P. Assessing the impact of climate change on building heating and cooling energy demand in Canada. Renew Sustain Energy Rev. 2020 Apr 1;121:109681.
- Jeong DI, Sushama L, Diro GT, Khaliq MN, Beltrami H, Caya D. Projected changes to high-temperature events for Canada based on a regional climate model ensemble. Clim Dyn. 2016 May 1;46(9):3163–80.
- 17. Hamlet AF, Byun K, Robeson SM, Widhalm M, Baldwin M. Impacts of climate change on the state of Indiana: ensemble future projections based on statistical downscaling. Clim Change. 2020 Dec 1;163(4):1881–95.
- Goldmann K, Wessel J. Some people feel the rain, others just get wet: An analysis of regional differences in the effects of weather on cycling. Res Transp Bus Manag. 2021 Sep;40:100541.
- 19. Friss E. The Cycling City: Bicycles and Urban America in the 1890s. In: The Cycling City [Internet]. University of Chicago Press; 2015 [cited 2023 Apr 14]. Available https://www.degruyter.com/document/doi/10.7208/9780226211077/html
- 20. Pucher J, Jacobsen PL, Buehler R, Garrard J, Rissel C, Bauman A, et al. City Cycling [Internet]. Cambridge, UNITED STATES: MIT Press; 2012 [cited 2023 Apr 20]. Available from: http://ebookcentral.proquest.com/lib/mcgill/detail.action?docID=3339528
- 21. Götschi T, Garrard J, Giles-Corti B. Cycling as a Part of Daily Life: A Review of Health Perspectives. Transp Rev. 2016 Jan 2;36(1):45–71.
- 22. Government of Canada SC. Cycling in Canada [Internet]. 2017 [cited 2023 Apr 15]. Available from: https://www150.statcan.gc.ca/n1/pub/82-003x/2017004/article/14788-eng.htm
- 23. Ellison RB, Greaves S. Travel Time Competitiveness of Cycling in Sydney, Australia. Transp Res Rec. 2011 Jan 1;2247(1):99–108.
- 24. Pucher J, Buehler R. Making Cycling Irresistible: Lessons from The Netherlands, Denmark and Germany. Transp Rev. 2008 Jul 1;28(4):495–528.

- 25. Gössling S, Choi A, Dekker K, Metzler D. The Social Cost of Automobility, Cycling and Walking in the European Union. Ecol Econ. 2019 Apr 1;158:65–74.
- 26. de HJJ, Boogaard H, Nijland H, Hoek G. Do the Health Benefits of Cycling Outweigh the Risks? Environ Health Perspect. 2010 Aug;118(8):1109–16.
- 27. Ma L, Ye R. Utilitarian Bicycling and Mental Wellbeing. J Am Plann Assoc. 2022 Apr 3;88(2):262–76.
- Eve M, Cornet Y, Munshi T. Measuring road space consumption by transport modes: Toward a standard spatial efficiency assessment method and an application to the development scenarios of Rajkot City, India. J Transp Land Use. 2020 Nov 29;13:651–69.
- 29. Drut M. Spatial issues revisited: The role of shared transportation modes. Transp Policy. 2018 Aug 1;66:85–95.
- 30. Gössling S, Choi AS. Transport transitions in Copenhagen: Comparing the cost of cars and bicycles. Ecol Econ. 2015 May 1;113:106–13.
- 31. Hamilton TL, Wichman CJ. Bicycle infrastructure and traffic congestion: Evidence from DC's Capital Bikeshare. J Environ Econ Manag. 2018 Jan 1;87:72–93.
- 32. Bassett DR, Pucher J, Buehler R, Thompson DL, Crouter SE. Walking, Cycling, and Obesity Rates in Europe, North America, and Australia. J Phys Act Health. 2008 Nov 1;5(6):795–814.
- 33. Gotschi T. Costs and Benefits of Bicycling Investments in Portland, Oregon. J Phys Act Health. 2011 Jan;8(s1):S49–58.
- 34. Sælensminde K. Cost-benefit analyses of walking and cycling track networks taking into account insecurity, health effects and external costs of motorized traffic. Transp Res Part Policy Pract. 2004 Oct 1;38(8):593–606.
- 35. Lamu AN, Jbaily A, Verguet S, Robberstad B, Norheim OF. Is cycle network expansion cost-effective? A health economic evaluation of cycling in Oslo. BMC Public Health. 2020 Dec 7;20(1):1869.
- 36. Boschmann EE, Kwan MP. Toward Socially Sustainable Urban Transportation: Progress and Potentials. Int J Sustain Transp. 2008 Mar 17;2(3):138–57.
- 37. Satterthwaite DD Jane Bicknell, David, editor. Adapting Cities to Climate Change: Understanding and Addressing the Development Challenges. London: Routledge; 2009. 424 p.

- 38. Mi Z, Guan D, Liu Z, Liu J, Viguié V, Fromer N, et al. Cities: The core of climate change mitigation. J Clean Prod. 2019 Jan 10;207:582–9.
- 39. Chapman L. Transport and climate change: a review. J Transp Geogr. 2007 Sep 1;15(5):354–67.
- 40. Xia T, Nitschke M, Zhang Y, Shah P, Crabb S, Hansen A. Traffic-related air pollution and health co-benefits of alternative transport in Adelaide, South Australia. Environ Int. 2015 Jan 1;74:281–90.
- 41. Kou Z, Wang X, Chiu SF (Anthony), Cai H. Quantifying greenhouse gas emissions reduction from bike share systems: a model considering real-world trips and transportation mode choice patterns. Resour Conserv Recycl. 2020 Feb 1;153:104534.
- 42. McQueen M, MacArthur J, Cherry C. The E-Bike Potential: Estimating the Effect of E-Bikes on Person Miles Travelled and Greenhouse Gas Emissions. TREC Final Rep [Internet]. 2019 May 1; Available from: https://pdxscholar.library.pdx.edu/trec_reports/187
- 43. Chen J, Zhou D, Zhao Y, Wu B, Wu T. Life cycle carbon dioxide emissions of bike sharing in China: Production, operation, and recycling. Resour Conserv Recycl. 2020 Nov 1;162:105011.
- 44. D'Almeida L, Rye T, Pomponi F. Emissions assessment of bike sharing schemes: The case of Just Eat Cycles in Edinburgh, UK. Sustain Cities Soc. 2021 Aug 1;71:103012.
- 45. Steinman L, Doescher M, Levinger D, Perry C, Carter L, Eyler A, et al. Master Plans for Pedestrian and Bicycle Transportation: Community Characteristics. J Phys Act Health. 2010 Jan 1;7(s1):S60–6.
- 46. Fishman E. Cycling as transport. Transp Rev. 2016 Jan 2;36(1):1–8.
- 47. Pucher J, Buehler R. Sustainable Transport in Canadian Cities: Cycling Trends and Policies. Berkeley Plan J [Internet]. 2006 [cited 2023 May 23];19(1). Available from: https://escholarship.org/uc/item/0rr0t06s
- 48. Useche SA, Montoro L, Sanmartin J, Alonso F. Healthy but risky: A descriptive study on cyclists' encouraging and discouraging factors for using bicycles, habits and safety outcomes. Transp Res Part F Traffic Psychol Behav. 2019 Apr 1;62:587–98.
- 49. Böcker L, Dijst M, Faber J. Weather, transport mode choices and emotional travel experiences. Transp Res Part Policy Pract. 2016 Dec 1;94:360–73.

- 50. Nosal T, Miranda-Moreno LF, Krstulic Z. Incorporating Weather: Comparative Analysis of Annual Average Daily Bicyclist Traffic Estimation Methods. Transp Res Rec J Transp Res Board. 2014 Jan;2468(1):100–10.
- 51. Böcker L, Dijst M, Prillwitz J. Impact of Everyday Weather on Individual Daily Travel Behaviours in Perspective: A Literature Review. Transp Rev. 2013 Jan 1;33(1):71–91.
- 52. Heinen E, van Wee B, Maat K. Commuting by Bicycle: An Overview of the Literature. Transp Rev. 2010 Jan 1;30(1):59–96.
- 53. El-Assi W, Salah Mahmoud M, Nurul Habib K. Effects of built environment and weather on bike sharing demand: a station level analysis of commercial bike sharing in Toronto. Transportation. 2017 May 1;44(3):589–613.
- 54. Miranda-Moreno LF, Nosal T. Weather or Not to Cycle: Temporal Trends and Impact of Weather on Cycling in an Urban Environment. Transp Res Rec J Transp Res Board. 2011 Jan;2247(1):42–52.
- 55. Pörtner HO, Roberts DC, Tignor MMB, Poloczanska ES, Mintenbeck K, Alegría A, et al., editors. Summary for policymakers. In: Climate Change 2022: Impacts, Adaptation and Vulnerability Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press; 2022.
- 56. Committee on Climate Change and U.S. Transportation, Transportation Research Board, Division on Earth and Life Studies, National Research Council. Potential Impacts of Climate Change on U.S. Transportation: Special Report 290 [Internet]. Washington, D.C.: Transportation Research Board; 2008 [cited 2023 May 31]. Available from: https://www.nap.edu/catalog/12179
- 57. Vincent LA, Zhang X, Mekis É, Wan H, Bush EJ. Changes in Canada's Climate: Trends in Indices Based on Daily Temperature and Precipitation Data. Atmosphere-Ocean. 2018 Oct 20;56(5):332–49.
- 58. Canada's Changing Climate Report [Internet]. [cited 2023 Apr 15]. Available from: https://changingclimate.ca/CCCR2019/
- 59. Koetse MJ, Rietveld P. The impact of climate change and weather on transport: An overview of empirical findings. Transp Res Part Transp Environ. 2009 May 1;14(3):205–21.
- 60. Wang T, Qu Z, Yang Z, Nichol T, Clarke G, Ge YE. Climate change research on transportation systems: Climate risks, adaptation and planning. Transp Res Part Transp Environ. 2020 Nov 1;88:102553.

- 61. Ji T, Yao Y, Dou Y, Deng S, Yu S, Zhu Y, et al. The Impact of Climate Change on Urban Transportation Resilience to Compound Extreme Events. Sustainability. 2022 Jan;14(7):3880.
- 62. Buehler R, Pucher J. COVID-19 Impacts on Cycling, 2019–2020. Transp Rev. 2021 Jul 4;41(4):393–400.
- 63. Green S, Sakuls P, Levitt S. Cycling for health: Improving health and mitigating the climate crisis. Can Fam Physician. 2021 Oct;67(10):739–42.
- 64. Teixeira JF, Lopes M. The link between bike sharing and subway use during the COVID-19 pandemic: The case-study of New York's Citi Bike. Transp Res Interdiscip Perspect. 2020 Jul 1;6:100166.
- 65. Qiao H. Does COVID-19 Mark a New Era for Bike Share? A Study on Indego Bike Share Usage Before and After the COVID-19 Outbreak. University of Pennsilvania; 2021.
- 66. Hu S, Xiong C, Liu Z, Zhang L. Examining spatiotemporal changing patterns of bike-sharing usage during COVID-19 pandemic. J Transp Geogr. 2021;91(102997).
- 67. Wang H, Noland RB. Bikeshare and subway ridership changes during the COVID-19 pandemic in New York City. Transp Policy. 2021 Jun 1;106:262–70.
- 68. Orvin MM, Fatmi MR, Chowdhury S. Taking another look at cycling demand modeling: A comparison between two cities in Canada and New Zealand. J Transp Geogr. 2021 Dec;97:103220.
- 69. Puchades VM, Pietrantoni L, Fraboni F, De Angelis M, Prati G. Unsafe cycling behaviours and near crashes among Italian cyclists. Int J Inj Contr Saf Promot. 2018 Jan 2;25(1):70–7.
- 70. Albuquerque V, Andrade F, Ferreira J, Dias M, Bacao F. Bike-sharing mobility patterns: a data-driven analysis for the city of Lisbon. EAI Endorsed Trans Smart Cities. 2018 Jul 13;169580.
- 71. Chibwe J, Heydari S, Faghih Imani A, Scurtu A. An exploratory analysis of the trend in the demand for the London bike-sharing system: From London Olympics to Covid-19 pandemic. Sustain Cities Soc. 2021 Jun 1;69:102871.
- 72. Bean R, Pojani D, Corcoran J. How does weather affect bikeshare use? A comparative analysis of forty cities across climate zones. J Transp Geogr. 2021 Jul 1;95:103155.

- 73. Bergantino AS, Intini M, Tangari L. Influencing factors for potential bike-sharing users: an empirical analysis during the COVID-19 pandemic. Res Transp Econ. 2021 May 1;86:101028.
- 74. Gebhart K, Noland RB. The impact of weather conditions on bikeshare trips in Washington, DC. Transportation. 2014 Nov 1;41(6):1205–25.
- 75. Heydari S, Konstantinoudis G, Behsoodi AW. Effect of the COVID-19 pandemic on bike-sharing demand and hire time: Evidence from Santander Cycles in London. PLOS ONE. 2021 dic;16(12):e0260969.
- Nikiforiadis A, Ayfantopoulou G, Stamelou A. Assessing the Impact of COVID-19 on Bike-Sharing Usage: The Case of Thessaloniki, Greece. Sustainability. 2020 Jan;12(19):8215.
- 77. Choi J, Lee WD, Park WH, Kim C, Choi K, Joh CH. Analyzing changes in travel behavior in time and space using household travel surveys in Seoul Metropolitan Area over eight years. Travel Behav Soc. 2014 Jan;1(1):3–14.
- Pucher J, Buehler R, Merom D, Bauman A. Walking and Cycling in the United States, 2001–2009: Evidence From the National Household Travel Surveys. Am J Public Health. 2011 Dec;101(S1):S310–7.
- 79. Ledsham T, Zhang Y, Farber S, Hess P. Beyond downtown: factors influencing utilitarian and recreational cycling in a low-income suburb. Int J Sustain Transp. 2022 Jun 30;0(0):1–22.
- 80. Stinson MA, Porter CD, Proussaloglou KE, Calix R, Chu C. Modeling the Impacts of Bicycle Facilities on Work and Recreational Bike Trips in Los Angeles County, California. Transp Res Rec. 2014 Jan 1;2468(1):84–91.
- 81. Wessel J. Using weather forecasts to forecast whether bikes are used. Transp Res Part Policy Pract. 2020 Aug;138:537–59.
- 82. Heaney AK, Carrión D, Burkart K, Lesk C, Jack D. Climate Change and Physical Activity: Estimated Impacts of Ambient Temperatures on Bikeshare Usage in New York City. Environ Health Perspect. 2019 Mar;127(3):037002.
- Zhao J, Guo C, Zhang R, Guo D, Palmer M. Impacts of weather on cycling and walking on twin trails in Seattle. Transp Res Part Transp Environ. 2019 Dec;77:573–88.
- 84. Lanza K, Gohlke J, Wang S, Sheffield PE, Wilhelmi O. Climate change and physical activity: ambient temperature and urban trail use in Texas. Int J Biometeorol [Internet]. 2022 May 27 [cited 2022 Jun 1]; Available from: https://link.springer.com/10.1007/s00484-022-02302-5

- 85. Niemeier DA. Longitudinal Analysis of Bicycle Count Variability: Results and Modeling Implications. J Transp Eng. 1996 May;122(3):200–6.
- 86. Wessel J. Cycling in the dark the impact of Standard Time and Daylight Saving Time on bicycle ridership. PNAS Nexus. 2022 Mar 1;1(1):pgab006.
- 87. Zhao W. Modeling Seasonal and Weather Impacts on Cycling Count [Master of Urban Planning]. [Seattle, Washington]: University of Washington; 2016.
- 88. Nordback KL. ESTIMATING ANNUAL AVERAGE DAILY BICYCLISTS AND ANALYZING CYCLIST SAFETY AT URBAN INTERSECTIONS. [Denver, Colorado]: University of Colorado Denver; 2012.
- 89. Morton C. The demand for cycle sharing: Examining the links between weather conditions, air quality levels, and cycling demand for regular and casual users. J Transp Geogr. 2020 Oct;88:102854.
- 90. Motoaki Y. TIME-SERIES ANALYSIS OF BICYCLE COUNT DATA [Master of Science]. [Ithaca, New York]: Cornell University; 2015.
- 91. Shang WL, Chen J, Bi H, Sui Y, Chen Y, Yu H. Impacts of COVID-19 pandemic on user behaviors and environmental benefits of bike sharing: A big-data analysis. Appl Energy. 2021 Mar 1;285:116429.
- 92. Nguyen MH, Pojani D. The emergence of recreational cycling in Hanoi during the Covid-19 pandemic. J Transp Health. 2022 Mar 1;24:101332.
- 93. Miranda-Moreno LF, Nosal T, Schneider RJ, Proulx F. Classification of Bicycle Traffic Patterns in Five North American Cities. Transp Res Rec J Transp Res Board. 2013 Jan;2339(1):68–79.
- 94. Planning-Population-NYC Population Facts DCP [Internet]. [cited 2022 Jul 20]. Available from: https://www1.nyc.gov/site/planning/planning-level/nycpopulation/population-facts.page
- 95. US Department of Commerce N. NWS JetStream Climate [Internet]. NOAA's National Weather Service; [cited 2022 Jul 21]. Available from: https://www.weather.gov/jetstream/climates
- 96. PeopleForBikes [Internet]. [cited 2022 Jul 21]. PeopleForBikes 2022 City Ratings | Every ride. Every rider. Join us. Available from: https://cityratings.peopleforbikes.org/
- 97. Government of Canada SC. Profile table, Census Profile, 2021 Census of Population - Canada [Country];Montréal, Ville (V) [Census subdivision], Quebec [Internet]. 2022 [cited 2022 Jul 21]. Available from:

https://www12.statcan.gc.ca/census-recensement/2021/dppd/prof/index.cfm?Lang=E

- 98. Montréal V de. Cycling and bike paths [Internet]. [cited 2022 Jul 21]. Available from: https://montreal.ca/en/topics/cycling-and-bike-paths
- 99. Ritchie H, Mathieu E, Rodés-Guirao L, Appel C, Giattino C, Ortiz-Ospina E, et al. Coronavirus Pandemic (COVID-19). Our World Data [Internet]. 2020 Mar 5 [cited 2022 Jul 24]; Available from: https://ourworldindata.org/covid-stringency-index
- 100. Kaplan S, Wrzesinska DK, Prato CG. Psychosocial benefits and positive mood related to habitual bicycle use. Transp Res Part F Traffic Psychol Behav. 2019 Jul 1;64:342–52.
- Woodcock J, Tainio M, Cheshire J, O'Brien O, Goodman A. Health effects of the London bicycle sharing system: health impact modelling study. BMJ. 2014 Feb 13;348:g425.
- 102. Otero I, Nieuwenhuijsen MJ, Rojas-Rueda D. Health impacts of bike sharing systems in Europe. Environ Int. 2018 Jun 1;115:387–94.
- 103. Raustorp J, Koglin T. The potential for active commuting by bicycle and its possible effects on public health. J Transp Health. 2019 Jun 1;13:72–7.
- 104. Flusche D. Bicycling means business: The economic benefits of bicycle infrastructure. 27 S [Internet]. 2012 [cited 2022 Sep 29]; Available from: https://repository.difu.de/jspui/handle/difu/232014
- 105. Fishman E, Schepers P, Kamphuis CBM. Dutch Cycling: Quantifying the Health and Related Economic Benefits. Am J Public Health. 2015 Aug;105(8):e13–5.
- 106. Bullock C, Brereton F, Bailey S. The economic contribution of public bikeshare to the sustainability and efficient functioning of cities. Sustain Cities Soc. 2017 Jan 1;28:76–87.
- 107. Komanoff C, Roelofs C, Orcutt J, Ketcham B. Environmental benefits of bicycling and walking in the United States. Transp Res Rec. 1993;1405:7.
- 108. Zhang Y, Mi Z. Environmental benefits of bike sharing: A big data-based analysis. Appl Energy. 2018 Jun 15;220:296–301.
- 109. Johansson C, Lövenheim B, Schantz P, Wahlgren L, Almström P, Markstedt A, et al. Impacts on air pollution and health by changing commuting from car to bicycle. Sci Total Environ. 2017 Apr 15;584–585:55–63.

- Mrkajic V, Vukelic D, Mihajlov A. Reduction of CO2 emission and nonenvironmental co-benefits of bicycle infrastructure provision: the case of the University of Novi Sad, Serbia. Renew Sustain Energy Rev. 2015 Sep 1;49:232– 42.
- 111. Inside the New Plan to Make Paris '100% Cyclable.' Bloomberg.com [Internet]. 2021 Oct 22 [cited 2022 Oct 10]; Available from: https://www.bloomberg.com/news/articles/2021-10-22/how-paris-will-become-100-cyclable
- 112. Montréal Copenhagenize [Internet]. [cited 2022 Oct 10]. Available from: https://copenhagenizeindex.eu/cities/montreal
- 113. Saneinejad S, Roorda MJ, Kennedy C. Modelling the impact of weather conditions on active transportation travel behaviour. Transp Res Part Transp Environ. 2012 Mar 1;17(2):129–37.
- 114. Böcker L, Prillwitz J, Dijst M. Climate change impacts on mode choices and travelled distances: a comparison of present with 2050 weather conditions for the Randstad Holland. J Transp Geogr. 2013 Apr 1;28:176–85.
- 115. Edwards PN. History of climate modeling. WIREs Clim Change. 2011;2(1):128–39.
- AR6 Synthesis Report: Climate Change 2023 IPCC [Internet]. [cited 2023 Jun 16]. Available from: https://www.ipcc.ch/report/sixth-assessment-reportcycle/
- 117. Yeo MH, Nguyen HL, Nguyen VTV. Statistical tool for modeling of a daily precipitation process in the context of climate change. J Water Clim Change. 2019 Nov 29;12(1):18–31.
- 118. Nguyen VTV, Nguyen TD, Gachon P. On the linkage of large-scale climate variability with local characteristics of daily precipitation and temperature extremes: an evaluation of statistical downscaling methods. In: Advances in Geosciences [Internet]. World Scientific Publishing Company; 2006 [cited 2023 Jun 16]. p. 1–9. (Advances in Geosciences; vol. Volume 4). Available from: https://www.worldscientific.com/doi/abs/10.1142/9789812707208_0001
- 119. Sunyer MA, Madsen H, Ang PH. A comparison of different regional climate models and statistical downscaling methods for extreme rainfall estimation under climate change. Atmospheric Res. 2012 Jan 1;103:119–28.
- 120. Huber DG, Gulledge J. EXTREME WEATHER & CLIMATE CHANGE: UNDERSTANDING THE LINK AND MANAGING THE RISK. :13.

- 121. Wang X, Huang G, Liu J, Li Z, Zhao S. Ensemble Projections of Regional Climatic Changes over Ontario, Canada. J Clim. 2015 Sep 15;28(18):7327–46.
- 122. Cheng CS, Auld H, Li G, Klaassen J, Li Q. Possible impacts of climate change on freezing rain in south-central Canada using downscaled future climate scenarios. Nat Hazards Earth Syst Sci. 2007 Jan 22;7(1):71–87.
- 123. Gaitán CF. Effects of variance adjustment techniques and time-invariant transfer functions on heat wave duration indices and other metrics derived from downscaled time-series. Study case: Montreal, Canada. Nat Hazards. 2016 Sep 1;83(3):1661–81.
- 124. Dickau M, Matthews D, Guertin É, Seto D. Projections of declining outdoor skating availability in Montreal due to global warming. Environ Res Commun. 2020 May;2(5):051001.
- 125. Wadud Z. Cycling in a changed climate. J Transp Geogr. 2014 Feb 1;35:12– 20.
- 126. Galich A, Nieland S, Lenz B, Blechschmidt J. How Would We Cycle Today If We Had the Weather of Tomorrow? An Analysis of the Impact of Climate Change on Bicycle Traffic. Sustainability. 2021 Jan;13(18):10254.
- 127. Sharafi M. Station and City Level Modelling of Bike-Sharing System for Montreal [Internet] [M.E.]. [Ann Arbor, United States]; 2021 [cited 2022 Sep 27]. Available https://www.proquest.com/docview/2652595931/abstract/5C2645F5345A4EA2 PQ/1
- 128. Chan NW, Wichman CJ. Climate Change and Recreation: Evidence from North American Cycling. Environ Resour Econ. 2020 May 1;76(1):119–51.
- 129. Mathisen TA, Annema JA, Kroesen M. The effects of weather and climate change on cycling in northern Norway. Eur J Transp Infrastruct Res [Internet]. 2015 Apr 1 [cited 2022 Aug 23];15(2). Available from: https://journals.open.tudelft.nl/ejtir/article/view/3072
- 130. Ramesh AA, Nagisetti SP, Sridhar N, Avery K, Bein D. Station-level Demand Prediction for Bike-Sharing System. In: 2021 IEEE 11th Annual Computing and Communication Workshop and Conference (CCWC). 2021. p. 0916–21.
- Broucke SV, Piña LMV, Do TH, Deligiannis N. BRUBIKE: A Dataset of Bicycle Traffic and Weather Conditions for Predicting Cycling Flow. In: 2019 IEEE International Smart Cities Conference (ISC2). 2019. p. 432–7.

- 132. Lainjo B. Application of Machine Learning in Predicting the Number of Bike Share Riders. Int J Bus Manag Econ. 2022 Nov 2;3(4):339–70.
- 133. Cui Y, Lv W, Wang Q, Du B. Usage Demand Forecast and Quantity Recommendation for Urban Shared Bicycles. In: 2018 International Conference on Cyber-Enabled Distributed Computing and Knowledge Discovery (CyberC). 2018. p. 238–2388.
- 134. Sathishkumar EV, Park J, Cho Y. Using data mining techniques for bike sharing demand prediction in metropolitan city. Comput Commun. 2020 Mar 1;153:353–66.
- 135. Boufidis N, Nikiforiadis A, Chrysostomou K, Aifadopoulou G. Development of a station-level demand prediction and visualization tool to support bike-sharing systems' operators. Transp Res Procedia. 2020 Jan 1;47:51–8.
- 136. Adame Valenzuela E, Barban P, Beitel D, Miranda Moreno LF, Nguyen VTV. Analyzing the Behavior and Growth of Cycling in Four North American Cities Before, During, and After the COVID-19 Pandemic. Transp Res Rec. 2023 Mar 17;03611981231157396.
- 137. Peng Y, Nagata MH. An empirical overview of nonlinearity and overfitting in machine learning using COVID-19 data. Chaos Solitons Fractals. 2020 Oct;139:110055.
- 138. Chen T, Guestrin C. XGBoost: A Scalable Tree Boosting System. In: Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining [Internet]. San Francisco California USA: ACM; 2016 [cited 2023 Aug 14]. p. 785–94. Available from: https://dl.acm.org/doi/10.1145/2939672.2939785
- 139. IPCC, editor. Emissions scenarios: summary for policymakers; a special report of IPCC Working Group III\$Intergovernmental Panel on Climate Change. Intergovernmental Panel on Climate Change; 2000. 20 p. (IPCC special report).
- 140. Makido A. Examining How Weather and Outdoor Thermal Comfort Variability Affects Cycling Activity Using Big Data and Microclimate SImulations in New York City. 2019 Aug 30 [cited 2023 May 16]; Available from: https://ecommons.cornell.edu/handle/1813/67734
- 141. Emmanuel R, Loconsole A. Green infrastructure as an adaptation approach to tackling urban overheating in the Glasgow Clyde Valley Region, UK. Landsc Urban Plan. 2015 Jun 1;138:71–86.

- 142. Sarat AA, Eusuf MA. EFFECTS OF PAVEMENT SURFACE TEMPERATURE ON THE MODIFICATION OF URBAN THERMAL ENVIRONMENT. DIMENSI J Archit Built Environ. 2012;39(1):1–4.
- 143. Cheela VRS, John M, Biswas W, Sarker P. Combating Urban Heat Island Effect—A Review of Reflective Pavements and Tree Shading Strategies. Buildings. 2021 Mar;11(3):93.
- Ruefenacht L, Acero JA. Strategies for Cooling Singapore: A catalogue of 80+ measures to mitigate urban heat island and improve outdoor thermal comfort. 2017 Jun 1 [cited 2023 May 16]; Available from: https://www.researchcollection.ethz.ch/handle/20.500.11850/258216
- 145. Montréal V de. Montréal's bicycle network [Internet]. [cited 2023 Jun 29]. Available from: https://services.montreal.ca/en/maps/bicycle-path
- 146. walking-and-cycling-in-vancouver-2018-report-card.pdf [Internet]. [cited 2022 Jul 21]. Available from: https://vancouver.ca/files/cov/walking-and-cycling-invancouver-2018-report-card.pdf